

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Spitzer, Sonja; Shaikh, Mujaheed

Working Paper Health Misperception and Healthcare Utilisation among Older Europeans

Vienna Institute of Demography Working Papers, No. 01/2020

Provided in Cooperation with: Vienna Institute of Demography (VID), Austrian Academy of Sciences

Suggested Citation: Spitzer, Sonja; Shaikh, Mujaheed (2020) : Health Misperception and Healthcare Utilisation among Older Europeans, Vienna Institute of Demography Working Papers, No. 01/2020, Austrian Academy of Sciences (ÖAW), Vienna Institute of Demography (VID), Vienna, https://doi.org/10.1553/0x003cb422

This Version is available at: https://hdl.handle.net/10419/247158

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



AUSTRIAN

SCIENCES

VIENNA INSTITUTE OF DEMOGRAPHY WORKING PAPERS

01/2020

HEALTH MISPERCEPTION AND HEALTHCARE UTILISATION AMONG OLDER EUROPEANS

SONJA SPITZER AND MUJAHEED SHAIKH

Vienna Institute of Demography Austrian Academy of Sciences Vordere Zollamtsstraße 3 | 1030 Vienna, Austria vid@oeaw.ac.at | www.oeaw.ac.at/vid



Abstract

Health perception biases can have serious consequences on health. Despite their relevance, the role of such biases in determining healthcare utilisation is severely underexplored. Here we study the relationship between health misperception, doctor visits, and concomitant outof-pocket expenditures for the population 50+ in Europe. We conceptualise health misperception as arising from either overconfidence or underconfidence, where overconfidence is measured as overestimation of health and underconfidence is measured as underestimation of health. Comparing objective performance measures and their self-reported equivalents from the Survey of Health, Ageing and Retirement in Europe, we find that individuals who overestimate their health visit the doctor 14% less often than individuals who correctly assess their health, which is crucial for preventive care such as screenings. Lower healthcare utilisation is accompanied by lower out-of-pocket spending (38% less). In contrast, individuals who underestimate their health visit the doctor more often (28% more) and have higher out-of-pocket spending (17% more). We project that underestimating health of the population 50+ will cost the average European country Intl\$ 71 million in 2020 and Intl\$ 81 million by 2060. Country-specific estimates based on population and demographic projections show that countries such as Germany, Denmark and The Netherlands will experience significantly large costs of such misperception. The results are robust to several sensitivity tests and, more important, to various conceptualisations of the misperception measure.

Keywords

Healthcare utilisation, health perception, overconfidence and underconfidence, doctor visits, out-of-pocket expenditures, SHARE data.

Authors

Sonja Spitzer (corresponding author), Wittgenstein Centre for Demography and Global Human Capital (Univ. Vienna, IIASA, VID/ÖAW), Austria. Email: sonja.spitzer@iiasa.ac.at

Mujaheed Shaikh, Hertie School, Germany. Email: shaikh@hertie-school.org

Acknowledgements

We are very grateful to Daniela Weber, Miguel Sanchez-Romero, Michael Kuhn, Raf Van Gestel, Anne Goujon, Marcel Bilger, Elsa Fontainha, and Monika Oczkowska for their valuable input. Also, we thank the participants of the 16th National Conference on Health Economics in Lisbon, the participants of the 6th International Workshop on the Socio-Economics of Ageing in Lisbon, the participants of the Wittgenstein Centre Conference 2019 in Vienna, and the participants of the Annual Meeting of the Austrian Economic Association 2020 in Vienna for their comments.

Health Misperception and Healthcare Utilisation among Older Europeans

Sonja Spitzer and Mujaheed Shaikh

1 Introduction

Biased perception of one's own ability is a hallmark of human nature. The literature in psychology, economics, and evolutionary biology has repeatedly demonstrated this phenomenon. Zell & Krizan (2014) conducted a meta-synthesis across different scientific areas and concluded that people have only moderate knowledge of their ability. Johnson & Fowler (2011) presented an evolutionary model of one such bias, namely, overconfidence, and the conditions under which it prevails. Such biases have significant implications for education, labour market outcomes, savings, investment choices, and political decisions (Anderson et al. 2017, Ortoleva & Snowberg 2015, Reuben et al. 2017). They are particularly relevant for health, as they can directly affect risk for accident and injury (Preston & Harris 1965, Sakurai et al. 2013) and have serious long-lasting effects on wellbeing and mortality. Recent work in this domain shows that overconfidence is related to engagement in risky health behaviours (Arni et al. 2019).

Despite the relevance of biased perception for health, its role in healthcare seeking is largely unexplored. Here we study the relationship between misperception of one's own health and future healthcare utilisation and medical expenditures. We categorise misperception as arising from either overconfidence or underconfidence in one's own health. Following the literature in psychology, we measure overconfidence as the overestimation of one's actual health and measure underconfidence as the underestimation of one's actual health (Moore & Healy 2008). It is a priori ambiguous how over- or underconfidence might relate to health-care use. On the one hand, individuals who overestimate their health may be less likely to visit the doctor when necessary, seek medical attention, or receive timely screenings because they believe their health is perfect. These individuals might also engage in more physical activity, which decreases healthcare utilisation (Rocca et al. 2015). On the other hand, the same individuals might engage in activity or behaviour detrimental to health and thus end up in the hospital more often. For example, older individuals who overestimate their mobility are more prone to fall-induced injuries (Sakurai et al. 2013). Similarly, individuals who underestimate their health may overutilise healthcare services by seeking care and purchasing

relatively more medication when it is not necessary—at least in the short run. In the long run, however, they might need less care and use fewer services because of their frequent doctor visits and timely diagnoses. Assessing the relationship between health perception and healthcare utilisation thus remains an empirical task that we undertake in this study.

Measuring over- or underconfidence bias in health is anything but trivial. It requires a subjective health measure and its objective equivalent, the lack of which often dissuades researchers from engaging in such research. We use a novel indicator to measure over- and underconfidence that is derived from the objective performance measures in the Survey of Health, Ageing and Retirement (SHARE). We analyse differences between subjective and objective health based on individuals' self-reported and tested ability to stand up from a chair. Individuals who subjectively report being able to stand but objectively are unable to do so are classified as overconfident, whereas those who subjectively report being unable to stand but objectively are able to are classified as underconfident. Individuals who do not differ in their subjective report and objective assessment are classified as concordant. Prior research has shown the chair stand test to be a good predictor of overall health (Ferrer et al. 1999, Sainio et al. 2006, Pinheiro et al. 2016, Spitzer & Weber 2019). Our approach distinguishes our measure of overconfidence from overplacement and overprecision because we focus only on individual judgements of completing a task rather than on relative comparisons with others or the estimated accuracy of such judgements (Moore & Healy 2008).

To assess utilisation, we use self-reported data on the annual number of doctor visits, which includes emergency room visits and outpatient clinic visits. Using count models, a rich set of controls, and longitudinal data, we find that relative to individuals who achieve concordance (i.e., those who estimate their health accurately), individuals who underestimate their health visit the doctor more often (approximately two more visits per year). In contrast, individuals who overestimate their health visit the doctor less often. We also analyse concomitant out-of-pocket (OOP) expenditures via log-Gamma models and find that individuals who underestimate their health have higher expenses, whereas individuals who overestimate their health have higher expenses, whereas individuals who overestimate their health have lower expenses. Our results are not biased by other individual characteristics, such as education, age, employment, or marital status, nor are they a manifestation of the inverse relationship between healthcare utilisation and the estimation of one's health as already stated. The results are robust to different model specifications, estimation methods, and measures of health perception.

We use data from 15 European countries from the SHARE survey, which provides other

advantages besides a measure of confidence. First, the longitudinal nature of the survey allows us to assess the relationship between confidence today and healthcare utilisation in the next wave of the survey. Thus, an important source of bias in our estimates—reverse causality—is not a first-order issue in our analyses. Second, utilising health services is conditional on having access to such services; a fair comparison of utilisation requires no significant difference in accessibility among the entities being compared. Universal coverage in European countries ensures that everyone has a certain level of access to the health system, unlike in the United States (OECD & European Commission 2018). Finally, Europe is a policy-relevant setting because of its rapidly ageing population (Lutz et al. 2003, Eurostat 2019) and fiscal pressures to reduce expenditures and unnecessary care (Christensen et al. 2009, European Commission 2018).

To quantify the public expenditure associated with health misperception, we perform a back-of-the-envelope calculation of the costs of health misperception. We project that underestimating health will cost the average European country Intl\$ 71 million in 2020 and Intl\$ 81 million by 2060. Although overestimating health results in negative costs due to lower numbers of doctor visits, these are in the short run only. In the long run, overestimation may result in individuals skipping timely screening and preventive care and lead to worse health, resulting in higher healthcare expenditures.

The contribution of this study is twofold. First, we introduce and advance a measure of health misperception in the health economics literature. Our measure of over- and underconfidence is simple and easy to calculate and an accurate indicator of health status. The medical literature has shown the chair stand test to be strongly correlated with physical health (Ferrer et al. 1999, Sainio et al. 2006, Pinheiro et al. 2016). Moreover, it is regularly performed in other surveys, such as the English Longitudinal Study of Ageing, which provides the opportunity to study different settings and make subsequent comparative analyses between countries.

Second, we contribute to at least two strands of the literature in health economics. The literature has repeatedly shown that individuals frequently over- or underestimate their own health status (Bago d'Uva et al. 2008, Beaudoin & Desrichard 2011, Coman & Richardson 2006, Furnham 2001, Jürges 2007). In addition, health perception differs by sociodemographic characteristics such as age (Srisurapanont et al. 2017, Crossley & Kennedy 2001), gender (Schneider et al. 2012, Merrill et al. 1997), country of residence (Spitzer & Weber 2019, Capistrant et al. 2014, Jürges 2007), education (Bago d'Uva et al. 2008, Choi & Caw-

ley 2017), and race (Jackson et al. 2017). The difference between subjective and predicted survival probability affects healthcare utilisation (Bíró 2016*a*), and individuals with higher expected longevity are more likely to go for cancer screening (Picone et al. 2004), suggesting that health perception affects healthcare utilisation. Our paper contributes to this strand by directly studying over- and underconfidence in one's own health.

It also contributes to the literature on the determinants of healthcare use. In explaining variation in health expenditures and healthcare utilisation, this literature focuses on either the supply side (i.e., provider confidence and precision) (Baumann et al. 1991, Berner & Graber 2008, Cutler et al. 2013, Meyer et al. 2013) or easily observable demand characteristics (e.g., age, gender, income, social class, employment and education) (Bíró 2013, Cameron et al. 2010, Tavares & Zantomio 2017, Vallejo-Torres & Morris 2013, Van Doorslaer et al. 2004, Zhang et al. 2018). Our paper makes a novel contribution by extending this literature to assess a difficult-to-observe demand variable that has consistently been shown to affect health.

The remainder of this paper is structured as follows. In Section 2, we describe the data and variables. In Section 3, we introduce our methodology. Section 4 presents and discusses the results, Section 5 describes a range of robustness analyses, Section 6 provides estimates for the total public cost of health misperception, and Section 7 concludes the paper.

2 Data and Descriptive Statistics

We analyse the relationship between health misperception and healthcare utilisation based on SHARE, a representative cross-country panel study of noninstitutionalised individuals ages 50 and older as well as their younger spouses (Börsch-Supan et al. 2013).¹ The survey provides rich information on health, socioeconomic background, employment, and social networks based on about 380,000 interviews with around 140,000 individuals. It is particularly well suited for studying European countries, as the data are ex-ante harmonised. Also, because it focuses on older individuals, who generally have higher healthcare needs than the young, it is the ideal data source for our analyses. SHARE was previously used to anal-

¹This study uses data from SHARE Waves 1, 2, 4, 5, and 6 (DOIs: 10.6103/SHARE.w1.700, 10.6103/SHARE.w2.700, 10.6103/SHARE.w4.700, 10.6103/SHARE.w5.700, 10.6103/SHARE.w6.700). SHARE data collection has been funded by the European Commission through FP5 (QLK6-CT-2001-00360), FP6 (SHARE-I3: RII-CT-2006-062193, COMPARE: CIT5-CT-2005-028857, SHARELIFE: CIT4-CT-2006-028812), FP7 (SHARE-PREP: GA N211909, SHARE-LEAP: GA N227822, SHARE M4: GA N261982), and Horizon 2020 (SHARE-DEV3: GA N676536, SERISS: GA N654221) and by DG Employment, Social Affairs & Inclusion. Additional funding from the German Ministry of Education and Research, the Max Planck Society for the Advancement of Science, the U.S. National Institute on Aging (U01_AG09740-1352, P01_AG005842, P01_AG08291, P30_AG12815, R21_AG025169, Y1-AG-4553-01, IAG_BSR06-11, OGHA_04-064, HHSN271201300071C), and various national funding sources is gratefully acknowledged (see www.share-project.org).

yse healthcare utilisation by, among others, Bíró (2014), Bolin et al. (2009), Paccagnella et al. (2013), and Tavares & Zantomio (2017).

2.1 Sample Construction

The chair stand test, which we use to determine our measure of over- and underconfidence, is used only in SHARE Wave 2 (2006/2007) and Wave 5 (2013). Because we are assessing the relationship between this measure of confidence and healthcare utilisation in the next wave, our dependent variables, namely, annual number of doctor visits and concomitant OOP expenditures, are taken from the next waves, that is, Wave 4 (2010–2012)² and Wave 6 (2015) (Börsch-Supan 2019*b*,*c*). Hence, we treat the data as pooled cross-sections by matching individuals' misperception at Waves 2 and 5 (*w*) with their utilisation at Waves 4 and 6 (*w* + 1), respectively.

Our dependent variables are taken from wave w + 1, which is why we drop all observations that do not provide information on doctor visits at wave w + 1. This affects mostly respondents who participated in Wave 2 but not in the subsequent Wave 4 or respondents who participated in Wave 5 but not in the subsequent Wave 6. We also exclude all respondents younger than 50 years and all observations based on proxy respondents. Overall, this results in 58,897 observations from 15 European countries, namely, Austria, Belgium, Czechia, Denmark, Estonia, France, Germany, Italy, Luxembourg, The Netherlands, Poland, Slovenia, Spain, Sweden, and Switzerland. The sample for OOP payments is smaller (41,868 observations), as OOP payments were not captured in Wave 4 (Section 2.2.2).

Based on their results on the chair stand test, we categorise individuals into three groups: those who achieve concordance (i.e. subjectively report having no problem standing up from the chair and objectively are able to or subjectively report having problems standing up from the chair and objectively are not able to), those who are overconfident (i.e., overestimate their health; subjectively report being able to stand up but objectively are unable to), and those who are underconfident (i.e., underestimate their health; subjectively report being unable to stand up but objectively are able to). With concordance as the reference category, the sample is split into two groups: those who are overconfident and those who are underconfident. Further details are provided in Section 2.3.

For the main analysis, health misperception is based on the chair stand variables, because they are binary and therefore clearly indicate whether an individual is unimpaired or im-

²SHARE Wave 3 focuses on people's life histories and thus is not utilised in our analyses.

paired. For robustness, we use additional measures of health perception based on subjective cognition and walking ability and their objective counterparts. We therefore add more waves to the analyses for robustness (Section 5.5).

2.2 Outcome Variables

In line with the literature, we use the annual number of doctor visits as a proxy for healthcare utilisation (see Bago d'Uva & Jones 2009, Bíró 2016*b*, Bolin et al. 2009, Lugo-Palacios & Gannon 2017, Tavares & Zantomio 2017, Zhang et al. 2018, among others). By analysing this number, we are able to capture the effects of health perception on public expenditures, as doctor visits are frequently subsidised by the public. In addition, doctor visits are good indicators of healthcare seeking in general and preventive healthcare and screenings in particular. In addition to doctor visits, we analyse annual OOP payments for doctor visits, which allows us to analyse the effects of health perception on private healthcare expenses.

2.2.1 Annual Doctor Visits

The annual number of doctor visits, emergency room visits, and outpatient clinic visits is ascertained by answers to the following question: "Now please think about the last 12 months. About how many times in total have you seen or talked to a medical doctor or qualified/ registered nurse about your health? Please exclude dentist visits and hospital stays, but include emergency room or outpatient clinic visits." The survey question is phrased almost identically in Waves 4 and 6; however, the words "or qualified/registered nurse" are excluded in Wave 4. For this and other reasons, we run separate estimations for each wave as a sensitivity analysis (Section 5.4).

The number of doctor visits is top-coded at 98 visits per year. On average, individuals in our sample visit the doctor seven times per year. The median, however, is lower (five times), which demonstrates the variable's strong right-skewness (Table 2). Naturally, individuals who suffer from chronic diseases or activity limitations visit the doctor more frequently than healthy individuals; thus, the number of doctor visits also increases with age. Gender differences in doctor visits are clear: Women have more annual doctor visits than men. A socioeconomic gradient is also observed with respect to education: The number of doctor visits decreases as education increases. It is interesting that individuals with supplementary insurance have fewer doctor visits than those without, a finding quite contrary to the literature, which predicts moral hazard with supplemental insurance (Coulson et al. 1995, Buchmueller et al. 2004) (see Table A.1 in the Appendix).

2.2.2 OOP expenditures for doctor visits

If participants report that they have seen or talked to a doctor, they are asked, "Did you pay anything yourself for your doctor visits (in the last twelve months)? Please also include expenses for diagnostic exams, such as imaging or laboratory diagnostics." If they answer "yes", they are then asked, "Overall, how much did you pay yourself for your doctor visits (in the last twelve months), that is how much did you pay without getting reimbursed by (a health insurance/ your national health system/ a third party payer)?" The amount of OOP payments is based on the latter question; it is set to zero if the respondent did not visit a doctor at all or if he or she claims zero payments for doctor visits. All values are presented in Euros. Implausibly large values are set to missing, as suggested by SHARE (Jürges 2015). This affects 3,006 observations.

OOP payments are available in Wave 6 but not in Wave 4; thus, we assess the association between health perception at Wave 5 (w) and OOP expenditures at Wave 6 (w + 1) only. Consequently, the sample is smaller for analyses of OOP payments than those of doctor visits. Because potential deductibles include expenditures for not only doctor visits but also other healthcare services, such as dentist visits and hospital stays, we do not consider deductibles when calculating the OOP expenditures variable.

The mean OOP expenditure is 73 Euros per year. However, 61% of the participants have zero OOP payments at Wave 6; thus, the median is zero (Table 2). It is interesting that OOP payments do not increase with the number of chronic diseases or activity limitations, but educational attainment has a strong positive correlation with OOP expenditures. Furthermore, mean OOP payments vary substantially between countries: They are highest in Luxembourg, Switzerland, Italy, and Austria, reflecting differences in utilisation and/or costsharing mechanisms (Paccagnella et al. 2013) (see Table A.2 in the Appendix).

2.3 Explanatory Variable: Health Perception

Following the literature in psychology, our measure of misperception relates to the most common interpretation of over- and underconfidence, namely, over- and underestimating one's performance, actual ability, chance of success, or level of control (Moore & Healy 2008). Assuming an underlying true level of health, we group individuals according to their perception of their health status. More specifically, we differentiate among individuals who perceive their health status correctly (concordance), those who believe that they are healthier than they really are (overestimation), and those who believe that they are unhealthier than they really are (underestimation). The true level of health is proxied by objective perfor-



Figure 1: Survey question ascertaining subjective impairment (response category proportions in brackets)

Do vou think it would be safe	_	No (12,4%, N=7,325)	7	Respondent stood up without using arms	(98.0%, N=48,032)
for you to try to stand up from a	\leq	Voc (83.2% N=40.004)	$ \longrightarrow $	Respondent used arms to stand up	(0.8%, N=406)
chair without using your arms?		165 (03.2%, N=49,004)	\rightarrow	Test not completed	(1.1%, N=526)
		Other (4.4%, N=2,568)		Other	(0.1%, N=40)

Figure 2: Sequence of questions ascertaining objective impairment (response category proportions in brackets)

mance measures data based on physical performance measures. This objective information about the respondent's health is matched with the respondent's subjective assessment of his or her health, thus revealing whether that individual's beliefs are correct or not.

SHARE provides several objective performance measures that can be utilised as proxies for true health. The measure most suited to analysing differences between objective and subjective health is the ability to stand up from a chair, as this self-assessed variable relates directly to its tested equivalent. This measure has been used previously in Spitzer & Weber (2019). In additional analyses, we also observe the differences between subjective and objective cognition as well as subjective and objective walking ability (Section 5.5).

To evaluate subjective ability to get up from a chair, survey participants are asked whether they have difficulties getting up from a chair. Figure 1 provides the detailed survey question. Individuals are considered subjectively impaired if they report difficulties getting up from a chair and subjectively unimpaired if they do not. Overall, 17.0% of the survey participants in our sample are considered subjectively impaired. Both the impaired and unimpaired groups are then subjected to the objective assessment.

In the objective assessment, individuals are asked to physically stand up from a chair.³ The chair stand test is introduced with the interviewer saying, "The next test measures the strength and endurance in your legs. I would like you to fold your arms across your chest and sit so that your feet are on the floor; then stand up keeping your arms folded across your chest. Like this ..." The exact sequence of questions leading to the chair stand test is shown in Figure 2. Individuals are considered objectively unimpaired if they stand up

³It is important to note that the chair stand test in Wave 2 was only conducted among those younger than 76 years. Thus, the sample is younger than 76 for any country that participated only in Wave 2.

Table 1: Overview health perception categories

Subjectively	Objectively						
	Unimpaire	d	Impaired				
Unimpaired	Pos. concordance:	87.6%	Overestimating:	56.9%			
Impaired Total	Underestimating:	12.4% 100.0%	Neg. concordance:	43.1% 100.0%			

Note: No weights applied

without using their arms and objectively impaired if they are not able to stand up from the chair, if they have to use their arms to stand up, or if they think it is unsafe to try to stand up from the chair.

Following the subjective report of impairment (i.e., unimpaired or impaired) and the subsequent objective test, individuals can either achieve concordance, overestimate their own health, or underestimate their own health. If they subjectively report being unimpaired but are objectively impaired, they overestimate their health. Likewise, if they subjectively report being impaired but are objectively unimpaired, they underestimate their health. Although the categorisation of over- and underestimation is straightforward, the categorisation of concordance (i.e., accurate beliefs about their health status) requires further consideration. Given true (objective) health, it is important to distinguish between two types of concordance. Individuals with a poor health status (i.e., objectively impaired) are classified as "negative concordance" if they also subjectively report being impaired. Likewise, individuals with a good health status (i.e., objectively unimpaired) are classified as "positive concordance" if they also subjectively report being impaired. Likewise, individuals with a good health status (i.e., objectively unimpaired) are classified as "positive concordance" if they also subjectively report being unimpaired. The four health perception outcomes are shown in Table 1.

Distinguishing between the two types of concordance ensures that we use the appropriate reference category for over- and underestimation in regression analyses. Overestimation can only be measured in the group whose objective health is impaired yet who subjectively report being unimpaired. Therefore, an appropriate group of individuals to compare to are those who are also objectively impaired (i.e., negative concordance). Underestimation can only be measured in the group whose objective health is unimpaired yet subjectively report impaired. The appropriate comparator for these individuals is the group that is also objectively unimpaired. This separation of the concordance group also provides an important empirical advantage; it ensures that we compare like with like in terms of true initial health thereby ridding ourselves of an important source of endogeneity, namely variation in health that can determine utilisation.

	Ν	Mean	Std. Dev.	Min.	Max.	Median
Healthcare utilisation						
Annual number of doctor visits at w+1	58,764	7.332	9.423	0	98	5
Annual out-of-pocket expenditure for doctor visits at w+1	39,988	73.349	298.196	0	47,500	0
Health perception						
Positive concordance (1 = yes)	56,152	0.743	0.437	0	1	1
Underestimating $(1 = yes)$	56,152	0.101	0.302	0	1	0
Negative concordance $(1 = yes)$	56,152	0.060	0.237	0	1	0
Overestimating (1 = yes)	56,152	0.096	0.295	0	1	0
Impairment						
Subjective impairment (1 = impaired)	58,758	0.170	0.376	0	1	0
Objective impairment (1 = impaired)	56,157	0.156	0.363	0	1	0
Health variables						
Number of chronic diseases at w	58,702	1.145	1.217	0	10	1
Number of chronic diseases at w+1	58,754	1.207	1.231	0	9	1
Number of activity limitations at w	58,755	0.357	1.177	0	13	0
Number of activity limitations at w+1	58,752	0.490	1.451	0	13	0
Control variables						
Age (in number of years)	58,764	64.521	9.765	50	100	63
Gender (1 = female)	58,764	0.545	0.498	0	1	1
Low education $(1 = yes)$	57,979	0.430	0.495	0	1	0
Medium education $(1 = yes)$	57,979	0.369	0.483	0	1	0
High education $(1 = yes)$	57,979	0.201	0.401	0	1	0
Is retired $(1 = yes)$	58,471	0.509	0.500	0	1	1
Is married $(1 = yes)$	56,883	0.680	0.466	0	1	1
Household income (in Euros per year)	58,764	46,569.89	76,244.77	0	1,200,000	24,000
Health access (1 = difficult)	39,120	0.163	0.370	0	1	0

Note: Calibrated cross-sectional individual weights are applied. For more detailed cross-tabulations see Tables A.1 to A.3 in the Appendix.

As shown in Table 1, in the objectively impaired group, 57% overestimate their health status; in the unimpaired group, only 12% underestimate. The large number of people reporting overconfidence is not surprising, as it has been documented in psychology and evolutionary theory as being favoured by natural selection and providing adaptive gains. Individuals tend to be overconfident because it increases morale and ambition and may thus improve potential (Johnson & Fowler 2011). Furthermore, our sample consists of older people, among whom overconfidence is particularly prevalent (Idler 1993, Spitzer & Weber 2019) and is seen as a resilience strategy to maintain a positive self-image (Brandtstädter & Greve 1994).

2.4 Additional Control Variables

We control for a range of variables that might otherwise confound our results. Summary statistics for these control variables are provided in Table 2, and cross-tabulations of control variables, doctor visits, health expenditures, and health perception are provided in Tables A.1 to A.3 in the Appendix. Most important, we control for other health factors at wave

w. In particular, we include the number of chronic diseases and the number of limitations in instrumental activities of daily living (IADLs) in our model. Chronic conditions that we consider are heart problems, high blood pressure or hypertension, high blood cholesterol, stroke or cerebral vascular disease, diabetes, chronic lung diseases, cancer, stomach or duodenal ulcer, Parkinson's disease, cataracts, hip fractures, other fractures, and Alzheimer's disease. A total of 35% of the sample have no chronic diseases at wave *w*; the weighted mean is 1.2 diseases. IADLs that we consider are difficulties dressing, walking across a room, bathing or showering, eating and cutting up food, getting in or out of bed, using the toilet, using a map, preparing a hot meal, shopping for groceries, making a telephone call, taking medications, doing work around the house or garden, and managing money. A total of 81% of the sample have no IADLs at wave *w*; the weighted mean is 0.5 IADLs. We only consider chronic diseases and IADLs that are included in both Wave 2 and Wave 5.

We also control for sociodemographic characteristics, as they are expected to influence health perception as well as healthcare utilisation (Avitabile et al. 2011, Lange 2011). In particular, we include age and age squared, gender, and educational attainment according to the International Standard Classification of Education (Eurostat 2018). Because pensioners appear to have higher healthcare utilisation (Bíró 2016*b*, Zhang et al. 2018), we also consider whether an individual is retired as opposed to all other employment options (employed, self-employed, unemployed, permanently sick or disabled, homemaker, other). Also, we control for whether the survey participant is married or in a registered partnership as opposed to never married, divorced, or widowed.

The effects of economic resources on healthcare utilisation are considered via equivalised household income. Because there are many missing values for household income in SHARE, the data set comes with two additional imputed variables. We use one of these imputed variables in our model and conduct a robustness analysis with the other (Section 5.4). We equivalise household income by using the square root scale, in which household income is divided by the square root of household size. Using the Organisation for Economic Cooperation and Development equivalence scale is not feasible, as children cannot be identified unambiguously. Furthermore, we use a cube root transformation to normalise the skewed income distribution (Cox 2011). Standard log normalisation is not feasible because of the substantial number of zero values.⁴ We run a robustness analysis in which we use equivalised household income that was not normalised (Section 5.4).

⁴Results are robust to dropping observations with zero values in household income.

3 Method

Ideally, we would randomly assign health perception to individuals to elicit causal effects of (mis)perception on healthcare utilisation and expenditures. In the absence of such random assignment, we rely on the panel dimension of the SHARE survey and control for a rich set of variables to account for confounding effects and bias due to reverse causation. Health perception is expected to affect healthcare utilisation, but the opposite mechanism, that healthcare utilisation precedes health perception, appears plausible too. For example, individuals who frequently visit the doctor are more likely to achieve concordance, as they receive more information about their health status. To overcome potential endogeneity, we analyse the effects of current health perception (wave w) on future healthcare utilisation (wave w + 1).

The main outcome variable—annual doctor visits—is strongly skewed to the right, yet without severe mass at zero. To accommodate this, we use a negative binomial model with mean dispersion, which is used frequently in the healthcare literature. We refrain from using a simple Poisson model, as the variance in the outcome variable is much larger than its mean. However, we perform robustness analyses using different models (Section 5.4). Thus, the number of doctor visits of individual *i* at wave w + 1 (DOCTOR_{*i*,*w*+1}) is assumed to follow a Poisson distribution but with a negative binomial specification for which each individual unit has a separate, Gamma-distributed mean. More specifically,

$$\text{DOCTOR}_{i,w+1} \sim \text{Poisson}(\mu_{i,w+1}),$$
 (1)

where

$$\mu_{i,w+1} = \exp(\beta \times \text{HEALTH PERCEPTION}_{i,w} + \gamma \times \text{HEALTH}_{i,w} + \delta \times X_{i,w} + \nu_i), \quad (2)$$

and

$$\exp(\nu_i) \sim \operatorname{Gamma}(1/\alpha, \alpha)$$
 (3)

HEALTH PERCEPTION is a binary variable that indicates whether individual i achieves concordance or misperceives his or her health at wave w. The vector HEALTH includes two variables, namely the number of chronic diseases in period w as well as the number of

IADLs in period *w* (thus, in the same period as health perception). The vector of control variables $X_{i,w}$ includes age and age squared, the individual's gender, educational attainment, household income, and control dummies for the survey wave as well as for the country of residence. The terms β , γ , and δ represent coefficients.

As discussed earlier, the sample is split into individuals who are overconfident (i.e., overestimate their health status) and individuals who are underconfident (i.e., underestimate it). The regression coefficients are therefore interpreted relative to those who estimate their health correctly (i.e., achieve concordance). For heterogeneity analyses, we further split the sample by gender, country, and number of chronic diseases.

When analysing the effects of health perception on OOP expenditures, we use a nonlinear model with a log link and Gamma family instead of the negative binomial model to account for the continuous nature of the outcome variable as well as for the excess zeros (Deb & Norton 2018). The specification of the variables included, however, remains identical to that described in Equation 2.

A total of 32% of the survey respondents participate in Waves 2, 4, 5, and 6, which allows us to analyse how health perception varies between Wave 2 and Wave 5 for these observations. For the majority (75%), health perception does not vary with age. If health perception changes, the most common changes are from underestimating to concordance (7.6%), from concordance to overestimating (6.5%), and from concordance to underestimating (5.6%). Because there is not enough variation in health perception within individuals, we refrain from using individual fixed effects in our analyses.

In Section 5.5, we explore whether our results are robust to different specifications of health perception. In particular, we estimate Equation 2 using cognition and the ability to walk as bases for the health perception variable.

4 **Results**

We first present the main results for the link between health misperception and healthcare utilisation and expenditures. We then examine heterogeneity in the relationships and present the results of important robustness analyses. Finally, we provide results for alternative measures of health perception bias.

4.1 Main Results

	(1)	(2)	(3)	(4)
	Objectively	Objectively	Objectively	Objectively
	Unimpaired	Unimpaired	Impaired	Impaired
	Doctor visits	OOP	Doctor visits	OOP
Health perception (ref.: concordance)				
Underestimating	0.244***	0.166^{*}		
U U	(0.018)	(0.077)		
Overestimating			-0.146***	-0.378**
Ũ			(0.029)	(0.138)
Chronic diseases	0.181***	0.118***	0.149***	0.118^{*}
	(0.005)	(0.031)	(0.009)	(0.055)
Activity limitations	0.096***	0.088	0.048^{***}	0.031
	(0.010)	(0.045)	(0.007)	(0.030)
Age	-0.001	0.082	0.021	0.165^{*}
	(0.011)	(0.058)	(0.018)	(0.082)
Age squared	0.000	-0.001	-0.000	-0.001*
	(0.000)	(0.000)	(0.000)	(0.001)
Woman	0.042***	0.125^{*}	0.014	0.422***
	(0.013)	(0.060)	(0.028)	(0.117)
Educ. group (ref.: low)				
Medium	0.006	0.419***	-0.006	0.112
	(0.016)	(0.072)	(0.033)	(0.126)
High	-0.003	0.881^{***}	-0.087*	0.544^{***}
	(0.018)	(0.093)	(0.042)	(0.152)
Retired	0.029	-0.031	0.014	0.398
	(0.017)	(0.104)	(0.033)	(0.228)
Married	-0.034*	0.029	0.019	0.405***
	(0.015)	(0.072)	(0.030)	(0.115)
Equiv. hh income (cube root)	-0.001	0.014***	-0.001	0.006
	(0.001)	(0.003)	(0.002)	(0.007)
Wave 5	-0.089***		-0.046	
	(0.015)		(0.038)	
Constant	1.507***	0.626	1.434^{*}	-1.232
	(0.356)	(1.963)	(0.646)	(2.828)
Country dummies	Yes	Yes	Yes	Yes
N	46,067	32,564	7,801	5,603
AIC	260,957	297,483	50,545	50,035
BIC	261,202	297,684	50,740	50,194

Table 3: Annual number of doctor visits and OOP expenditures for doctor visits at w+1

Note: "Unimpaired" refers to the sample that is objectively unimpaired, i.e. able to stand up from the chair and "Impaired" refers to the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w+1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. The dependent variable "OOP" is based on annual out-of-pocket payments for doctor visits at wave w+1, i.e. Wave 6. All explanatory variables are taken from wave w, i.e. The coefficients are estimated based on a generalised linear model model with log link and a Gamma family. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

4.1.1 Healthcare Utilisation

Healthcare utilisation is measured by the annual number of doctor visits. Table A.1 of the Appendix shows that overall, individuals who overestimate their health have fewer doctor

visits (8.6 visits) compared to their reference group (i.e., negative concordance = 11.9 visits). Similarly, those who underestimate their health have significantly more doctor visits in a year (10.1 visits) compared to their relevant reference group (i.e., positive concordance = 6.2 visits).

The table also shows cross-tabulations by other characteristics of the sample. Using number of chronic conditions and activity limitations as proxies for doctor visits provides two important insights. First, we find that as illness increases, so does the number of doctor visits irrespective of the category of perception bias. Second, at every level of illness, individuals who underestimate their health visit the doctor more often than those who are concordant. Similarly, overall, at almost each level of health (barring a few exceptions), individuals who overestimate their health visit the doctor less often. This shows that despite the same underlying health status, there is variation in doctor visits by health misperception category. Starkly similar results are observed for increasing age.

Although the picture is somewhat mixed across education categories, we observe fewer doctor visits for overestimators relative to their concordant counterparts at every level of education. Similarly, underestimators have higher healthcare utilisation than their concordant counterparts at each level of education. Accessibility to health professionals strongly determines health access; the pattern of utilisation across this variable by our misperception category remains the same as before: Overestimation shows fewer doctor visits, and underestimation shows more visits. Similar results are observed by supplementary insurance status.

These descriptive findings show that despite conditioning on individual characteristics, there is clear variation in healthcare utilisation in the form of doctor visits among the different health perception categories. In the regression analyses, we control for these and other variables such as country dummies. Table 3 shows the regression results. Columns 1 and 3 show the results for the two groups (i.e., overestimators and underestimators categorised based on the objective health status as impaired or unimpaired). All coefficients are to be interpreted relative to the concordance category.

We find a strong and significant association between health misperception and healthcare utilisation. Individuals who underestimate their health visit the doctor 27.6% more often in the subsequent period than individuals who achieve concordance. Computing marginal effects at means shows that this results in approximately two additional doctor visits per year.

We also find a strong and significant link between overestimation and the annual number of doctor visits. Individuals who overestimate their health go to the doctor less often than those who achieve concordance. Overestimating health at wave w results in 13.6% fewer doctor visits at wave w + 1 compared to perceiving one's health correctly. The marginal effect at means of overestimating health on healthcare utilisation is approximately 1.3 fewer doctor visits per year.

The results for doctor visits in Table 3 are based on a negative binomial model with mean dispersion. Figures 3 and 4 show that this model has the best fit among a simple Poisson model, a negative binomial model with constant dispersion, and a zero-inflated Poisson model.

4.1.2 OOP Expenditures

Individuals who visit a doctor also report OOP expenses, if any, measured in Euros. Table A.2 shows descriptive cross-tabulations of OOP expenditures incurred by other individual characteristics and by misperception category. Although no consistent pattern in OOP spending emerges by the number of chronic conditions, number of activity limitations, or increasing age, women show slightly higher expenditures than men. A clear education gradient is also observed in OOP spending, with higher education relating positively to spending. Similarly, retired individuals spend more than those who are not retired, as do married individuals compared to single ones. It is not surprising that in general individuals with supplementary insurance spend slightly less than those without it, and those with difficulty accessing healthcare spend less than those with no difficulty. Certain countries, such as Switzerland, Luxembourg, and Italy, show exceptionally high OOP expenditures, whereas others, such as France, Denmark, and Eastern European countries, show much lower OOP expenditures, which partly reflects institutional differences in user charges and the coverage of certain services.

It is interesting that we observe similar patterns as for doctor visits across the misperception categories. At almost every level of chronic conditions, activity limitations, and increasing age, we find that those who underestimate their health have higher OOP spending than those who achieve concordance. The findings for overestimation are somewhat mixed. Underestimating men (women) have slightly lower (much higher) OOP spending than their concordant counterparts. Overestimating men (women) also have slightly lower (higher) OOP spending than their concordant counterparts. Although an education gradient can be seen for underestimators with higher OOP spending compared to the concordant group at

Figure 3: Count model comparison for the annual number of doctor visits in the unimpaired sample, i.e. able to stand up from the chair



Number of doctor visits

Note: Doctor visits are top coded at 98 visits by year. This figure shows only the first 30 doctor visits for better visualisation. Dark bars represent the empirically observed numbers of doctor visits and light bars represent the predicted values based on the respective count model.

Figure 4: Count model comparison for the annual number of doctor visits in the impaired sample, i.e. unable to stand up from the chair



Note: Doctor visits are top coded at 98 visits by year. This figure shows only the first 30 doctor visits for better visualisation. Dark bars represent the empirically observed numbers of doctor visits and light bars represent the predicted values based on the respective count model.

each level of education, the same is not observed consistently for overestimators in the case of lower spending. Although underestimators consistently spend more OOP across the other individual characteristics, mixed findings are seen for overestimators. Regression results controlling for these characteristics in Columns 2 and 4 of Table 3 show that individuals who underestimate their health have significantly higher OOP expenses. On average, expenditures are 16.6% higher for those who underestimate their health compared to those who achieve concordance. In contrast, individuals who overestimate their health spend 37.8% less in OOP expenditures per year relative to their concordant group.

The results for OOP payments are based on a log-Gamma model. According to Akaike's information criterion and the Bayesian information criterion, the log-Gamma model has a better fit than either a log-Gaussian model or a log-Poisson model.

4.2 Heterogeneity of effects

We assess the heterogeneity of our main results in several ways. In particular, we consider gender differences, country specificities, and differences by health status.

4.2.1 Gender Differences

The literature has shown differences in health perception by individual characteristics, most important by gender (Merrill et al. 1997, Schneider et al. 2012). Gender differences in effects of health beliefs on healthcare utilisation may partly explain the well-documented differences in healthcare seeking between men and women, as men tend to have lower healthcare use (Galdas et al. 2005, Mansfield et al. 2003, Schlichthorst et al. 2016). Thus, we assess whether the relationship between health (mis)perception and utilisation also differs between men and women. As noted earlier, Table A.1 shows that, overall, women have slightly more doctor visits annually compared to men; this is true also within the misperception category, but the difference is not large. Furthermore, both under- and overestimating men and women have more doctor visits relative to their respective concordant comparators. In the case of OOP expenditures, however, whereas both under- and overestimating women have higher spending relative to their concordant group, both under- and overestimating men have higher spending relative to their concordant group. However, under- and overestimating men have lower spending relative to their concordant group. However, under- and overestimating men

Regression analyses by gender reveal that the association between health misperception and the annual number of doctor visits is slightly larger in magnitude for men than for women (Table 4). Marginal effects at means show that men who underestimate their health visit the doctor an additional 1.8 times compared to men who achieve concordance. For women, the difference is an additional 1.5 doctor visits. Men who overestimate their health have 1.5 fewer annual doctor visits compared to men who achieve concordance. For women, it is 1.3 fewer visits. A Wald test, however, reveals that the coefficients for women and men are not statistically different from each other.

4.2.2 Country Specificity

Differences in reporting behaviour by country are well documented (Capistrant et al. 2014, Jürges 2007, Spitzer & Weber 2019). To ensure that our findings are not driven by differential reporting due to cultural biases in reporting health or the oversampling of certain countries in the SHARE survey, we rerun our analyses for each country separately. By and large, we find similar results for all countries, for both under- and overestimation, with the exception of a few countries for which we do not find statistically significant results because of small sample sizes (see Tables A.4 and A.5 in the Appendix). However, it is worth noting that the magnitude of the coefficient for underestimation is much larger in certain countries, such as Denmark, Germany, and The Netherlands, than in others, perhaps reflecting differences in accessibility or other institutional differences in terms of, for example, user charges.

4.2.3 Differences by Health Status

Separating the underestimators and overestimators by objective health status allows us to overcome an important endogeneity concern related to initial health status that affects both health perception and healthcare utilisation. However, because we assess healthcare utilisation at w + 1, we also assess heterogeneity by health status at w + 1, which allows us to understand whether current health status drives differential utilisation in any way. Note that current utilisation at w + 1 will not drive health misperception because we assess misperception at w.

The descriptive statistics in Table A.3 in the Appendix indicate a slight decrease in concordance as the number of chronic diseases increases; however, this trend is far from obvious and might also be due to the correlation between health and age. To disentangle these effects, we run separate regressions for those individuals who do not have any chronic diseases at wave w + 1 (healthy) and those who report one or more chronic diseases at wave w + 1 (unhealthy). The results are reported in Table 5. Although health perception affects the doctor visits of impaired individuals with and without chronic diseases similarly, underestimation has a bigger effect on those without chronic diseases than on those with chronic diseases—this is confirmed by a Wald test. However, marginal effects reveal no substantial difference between the healthy and the unhealthy subsamples with respect to the relation-ship between over- or underestimation and doctor visits. Because we categorise based on

	(1) Objectively	(2) Objectively	(3) Objectively	(4) Objectively
	Unimpaired	Impaired	Unimpaired	Impaired
	Men	Men	Women	Women
	Doctor visits	Doctor visits	Doctor visits	Doctor visits
Health perception (ref. concordance)				
Underestimating	0 267***		0 229***	
onderestimating	(0.035)		(0.021)	
Overestimating	(0.000)	-0 139**	(0.021)	-0 136***
Overestillating		(0.048)		(0.034)
Chronic discasos	0 101***	0.168***	0 176***	0.1/3***
Childred diseases	(0.008)	(0.014)	(0.007)	(0.012)
Activity limitations	0.000)	(0.014)	0.007)	0.050***
Activity minitations	(0.000)	(0.012)	(0.000)	(0.008)
Δσρ	0.018	0.038	-0.008	0.014
nge	(0.017)	(0.030)	(0.013)	(0.012)
Age squared	-0.000	-0.000	0.000	-0.000
nge squared	(0,000)	(0,000)	(0,000)	(0,000)
Educ. group (ref.: low)	(0.000)	(0.000)	(0.000)	(0.000)
Medium	0.042	-0.074	-0.027	0.044
	(0.026)	(0.052)	(0.020)	(0.041)
High	0.017	-0.151*	-0.024	-0.055
8	(0.029)	(0.061)	(0.024)	(0.057)
Retired	0.031	0.075	0.026	-0.026
	(0.031)	(0.067)	(0.021)	(0.040)
Married	-0.045	0.032	-0.047*	-0.007
	(0.025)	(0.055)	(0.018)	(0.036)
Equiv. hh income (cube root)	-0.001	0.000	-0.001	-0.001
1	(0.001)	(0.002)	(0.001)	(0.002)
Wave 5	-0.116***	0.090	-0.066***	-0.124**
	(0.022)	(0.058)	(0.019)	(0.047)
Constant	0.654	0.604	1.983***	1.858*
	(0.583)	(1.059)	(0.441)	(0.788)
Country dummies	Yes	Yes	Yes	Yes
N	20,693	2,864	25,374	4,937
AIC	116,397	18,426	144,362	32,072
BIC	116,611	18,587	144,582	32,248

Table 4: Annual number of doctor visits at w + 1 by gender

Note: "Unimpaired" refers to the sample that is objectively unimpaired, i.e. able to stand up from the chair and "Impaired" refers to the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

health (in other words, fix health at the same level) we can conclude that the results are not driven by health differences: Both the healthy group's and the unhealthy group's healthcare utilisation is affected by their health perception in the same direction and to a similar magnitude.

	(1) Objectively	(2) Objectively	(3) Objectively	(4) Objectively
	Unimpaired	Unimpaired	Impaired	Impaired
	No chronic	Chronic	No chronic	Chronic
	dis. at w+1	dis. at w+1	dis. at w+1	dis. at w+1
Health perception (ref : concordance)				
Underestimating	0.351***	0 197***		
onderestantating	(0.042)	(0.020)		
Overestimating	(0.012)	(0.020)	-0.259***	-0.116***
			(0.066)	(0.031)
Chronic diseases	0.185***	0.093***	0.108***	0.106***
	(0.015)	(0.006)	(0.029)	(0.010)
Activity limitations	0.142***	0.087***	0.073***	0.045***
	(0.025)	(0.010)	(0.015)	(0.007)
Age	-0.027	-0.028*	-0.019	0.006
0	(0.018)	(0.012)	(0.039)	(0.020)
Age squared	0.000*	0.000**	0.000	-0.000
0 - 1	(0.000)	(0.000)	(0.000)	(0.000)
Woman	0.155***	0.008	0.116	-0.002
	(0.024)	(0.014)	(0.063)	(0.031)
Educ. group (ref.: low)	· · · ·	· · · ·	~ /	· · ·
Medium	0.022	0.018	0.023	-0.002
	(0.030)	(0.018)	(0.075)	(0.035)
High	0.066	-0.000	0.014	-0.071
0	(0.034)	(0.021)	(0.083)	(0.047)
Retired	0.023	0.020	0.094	-0.005
	(0.034)	(0.019)	(0.076)	(0.035)
Married	-0.033	-0.034*	-0.003	0.036
	(0.027)	(0.017)	(0.075)	(0.031)
Equiv. hh income (cube root)	0.002	-0.002*	0.004	-0.001
•	(0.001)	(0.001)	(0.003)	(0.002)
Wave 5	-0.046	-0.067***	-0.145	-0.001
	(0.027)	(0.017)	(0.074)	(0.042)
Constant	1.835**	2.821***	1.590	2.320***
	(0.614)	(0.419)	(1.348)	(0.704)
Country dummies	Yes	Yes	Yes	Yes
N	17.362	28.696	1.827	5.973
AIC	84,631	173,306	10.120	40.020
BIC	84,849	173,537	10,274	40,207

Table 5: Annual number of doctor visits at w + 1 by chronic diseases at w + 1

Note: "Unimpaired" refers to the sample that is objectively unimpaired, i.e. able to stand up from the chair and "Impaired" refers to the sample that is objectively impaired, i.e. unable to stand up from the chair. The sample "No chronic dis. at w + 1" includes those that have zero chronic diseases at wave w + 1, whereas "Chronic dis. at w + 1" refers to those that have one ore more chronic diseases at wave w + 1. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

5 Robustness Tests

We conduct a range of robustness analyses to observe whether our results are sensitive to model specifications and sample composition. These results are presented in Tables A.6 and A.7 in the Appendix along with the original model specification (Column 1).

5.1 Income

First, we utilise different income variables. We exchange the first imputed income variable provided by SHARE with the second imputed income variable (Column 2), and we use income that is not normalised with the cube root method but only equivalised (Column 3). These adjustments have no effects on the results. We also replace income with wealth (Column 4), and the results remain robust.

5.2 Wave Specific Analyses

Second, we separate the sample by survey wave to explore whether the slight change in the phrasing of the survey question about doctor visits in Wave 6 (Section 2.2.1), the restriction of the chair stand test to those younger than 76 years in Wave 2 (Section 2.3), or the different time gaps between w and w + 1 affect the results. The estimates in Table A.9 in the Appendix reveal that the effect of health misperception on healthcare utilisation is slightly stronger at Wave 5 than at Wave 2; however, the difference is not statistically significant according to a Wald test.

5.3 Response Reliability

Third, we exclude anyone diagnosed with Alzheimer's disease, dementia, or another serious memory impairment, as their survey answers might not be reliable (Column 5). The results remain robust, perhaps because the number of individuals observations with severe cognitive impairments in the survey is small.

5.4 Robustness to Further Controls

Finally, we conduct robustness analyses that are only possible for Wave 5, as they include variables only collected in this wave (see Table A.8 in the Appendix). First, we analyse whether differences in access to healthcare affect the number of doctor visits. For this, the household respondent is asked "How easy is it to get to your general practitioner or the nearest health center? Would you say it is very easy, easy, difficult or very difficult?" We dichotomise the variable by comparing the first two and the last two possible answers and add it to the model (Columns 2 and 5). The coefficients show, however, that the results do not depend on access to healthcare. Second, we investigate whether the results are robust to individuals purchasing supplementary health insurance. Although supplementary insur-

ance increases healthcare utilisation (Moreira & Barros Pita 2010, Paccagnella et al. 2013), we find no significant changes in our results (Column 6) when controlling for this variable.

5.5 Additional Measures of Health Perception

For the main analyses, health perception is operationalised based on tested and self-reported ability to stand up from a chair. We also analyse whether the results hold for other health dimensions, in particular health perception concerning cognition and walking ability.

5.5.1 Cognition

Similar to previous work, we use the difference between subjective and objective cognition as an additional measure of health perception (Spitzer & Weber 2019). Objective cognition is operationalised based on a memory test, which is conducted in Waves 4 to 6. In particular, individuals are asked to recall a list of 10 words in any order within a minute.

Subjective cognition is based on the question "How would you rate your memory at the present time?" which is answered on a Likert scale with the categories excellent, very good, good, fair, and poor. Because the subjective cognition variable has more than 80% missing values in Wave 6, we only utilise data from Waves 4 and 5. Hence, the estimates for cognition are based on a different sample. For the main results presented in Section 4, health perception from Waves 2 and 5 is matched with healthcare utilisation from Waves 4 and 6. For the results for cognition, health perception from Waves 5 and 6.

Defining cognitive impairment is not as straightforward as defining the ability to stand up from a chair. Whereas the chair stand variables are binary and therefore clearly indicate whether an individual is impaired, both the subjective and objective cognition variables are categorical. Thus, we rely on previous literature to define the threshold marking cognitive impairment. Participants are considered objectively impaired if they recall three words or fewer (Grodstein et al. 2001, Purser et al. 2005). In addition, in robustness analyses, individuals are considered impaired if they recall two words or fewer. Individuals are considered subjectively impaired if they report having a fair or poor memory (Gardner et al. 2017).

Tables 6 provides regression results for this new specification of health perception. The results confirm our earlier findings. Individuals who underestimate their cognitive ability at wave w are more likely to visit the doctor at wave w + 1 than individuals who achieve concordance between objective and subjective measures of memory. By contrast, survey participants who overestimate their health have fewer annual doctor visits than those who achieve concordance. Modifying the threshold for objective impairment from three to two words changes the magnitude of the coefficient for overestimation but not its sign. The magnitude of the coefficient for overestimation surface virtually identical.

5.5.2 Walking Ability

We also operationalise health perception based on walking ability. Objective walking ability is based on a walking speed test in which participants have to walk a distance of 2.5 m. Individuals are considered objectively impaired if their walking speed is 0.4 m per second or slower. This threshold is in line with the previous literature (Jürges 2007, Steel et al. 2003). Because the test is only conducted in Waves 1 and 2, the analysis is restricted to those waves (Börsch-Supan 2019*a*). The walking speed test is supposed to be conducted only for individuals older than 75 years. However, the data set includes information for those 75 and younger too. The variable has many missing values (~90%) and thus needs to be handled with caution.

Subjective walking impairment is based on the following question: "Please look at card [...]. We need to understand difficulties people may have with various activities because of a health or physical problem. Please tell me whether you have any difficulty doing each of the everyday activities on card [...]. Exclude any difficulties that you expect to last less than three months." Participants are coded as having subjectively impaired walking ability if they report difficulty walking 100 m.

When analysing health perception based on walking ability, we do not control for IADLs, as the ability to walk across a room is itself considered an IADL. Also, the second imputed income variable is used for this analysis, as the first one is not available in Wave 1. The robustness analysis in Section 5.4 shows, however, that both variables produce the same results.

Results for the effects of health perception on the annual number of doctor visits based on walking ability are provided in Table 7. The coefficients in Table 7 confirm once again that individuals who underestimate their health have more annual doctor visits than those who assess their health correctly. The results also show that those who overestimate their health have fewer doctor visits. Thus, our results are robust to different specifications of health perception.

	(1) Objectively Unimpaired 3 words	(2) Objectively Impaired 3 words	(3) Objectively Unimpaired 2 words	(4) Objectively Impaired 2 words
Health perception (ref.: concordance)				
Underestimating	0.079***			
Ū	(0.013)			
Overestimating		-0.067*		
-		(0.027)		
Underestimating			0.077***	
			(0.012)	
Overestimating				-0.145***
				(0.042)
Chronic diseases	0.194***	0.143***	0.189***	0.147***
	(0.004)	(0.009)	(0.004)	(0.014)
Activity limitations	0.114***	0.057***	0.106***	0.045***
	(0.005)	(0.007)	(0.005)	(0.009)
Age	0.013	0.042*	0.017*	0.055
	(0.009)	(0.021)	(0.008)	(0.031)
Age squared	-0.000	-0.000	-0.000	-0.000
T 4.7	(0.000)	(0.000)	(0.000)	(0.000)
vvoman	(0.078^{++++})	-0.003	(0.072^{-11})	-0.043
Educe around (ref. low)	(0.011)	(0.029)	(0.011)	(0.045)
Educ. group (ref.: low)	0.006	0.074*	0.010	0.012
Medium	(0.006)	(0.074)	(0.010)	(0.013)
Llich	(0.014)	(0.037)	0.013)	(0.036)
riigit	(0.013)	(0.010	(0.008)	(0.100)
Ratirad	(0.010)	(0.039)	(0.010)	0.100)
Kenred	(0.055)	(0.035)	(0.022)	(0.052)
Married	-0.025	-0.035	-0.028*	-0.030
Warred	(0.023)	(0.030)	(0.012)	(0.045)
Equiv. hh income (cube root)	-0.001*	-0.000	-0.001*	-0.003
	(0.001)	(0.002)	(0.001)	(0.002)
Wave 5	-0.037***	-0.054*	-0.036***	-0.100*
	(0.010)	(0.026)	(0.010)	(0.040)
Constant	1.086***	0.406	0.937**	0.245
	(0.302)	(0.750)	(0.287)	(1.114)
Control variables country	Yes	Yes	Yes	Yes
N	64,609	9,091	70,122	3.578
AIC	373.563	55.873	407.761	21.793
BIC	373,808	56,065	408,008	21,960

Table 6: Health perception based on cognition

Note: In columns 1 and 2, individuals are considered objectively impaired if they recall 3 words or less ("3 words"), while in columns 3 and 4 the cutoff is at 2 words or less ("2 words"). The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 5 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 4 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

	(1) Objectively Unimpaired	(2) Objectively Impaired
	Orimpuncu	impuncu
Health perception (ret.: concordance)	0.040**	
Underestimating	0.249**	
	(0.090)	0.000*
Overestimating		-0.328*
		(0.150)
Chronic diseases	0.129***	0.012
	(0.020)	(0.061)
Age	0.152*	0.337**
	(0.061)	(0.128)
Age squared	-0.001*	-0.002**
	(0.000)	(0.001)
Woman	-0.186**	-0.185
	(0.066)	(0.149)
Educ. group (ref.: low)		
Medium	0.189*	-0.041
	(0.075)	(0.203)
High	0.097	-0.235
0	(0.088)	(0.239)
Retired	-0.040	-0.202
	(0.085)	(0.178)
Married	-0.109	0.127
	(0.066)	(0.134)
Equiv. hh income (cube root) 2	-0.007	-0.023
1	(0.005)	(0.013)
Wave 2	-0.315*	-0.126
	(0.137)	(0.204)
Constant	-3.134	-9.194
	(2.197)	(4.759)
Control variables country	Yes	Yes
N	1,545	233
AIC	9,447	1,548
BIC	9,580	1.634

Table 7: Health perception based on walking ability

Note: The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 2 or Wave 4. All explanatory variables are taken from wave w, i.e. Wave 1 or Wave 2 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

6 The Total Public Cost of Health Misperception

We perform a back-of-the-envelope calculation to estimate additional health expenditures due to health misperception. The total cost of over- or underestimating health for the population 50+ in the respective country C_t in year t is calculated as follows:

$$C_{c,t} = d_c \times m_c \times f_c \times p_{c,t} \tag{4}$$

where d_c is the predicted cost per outpatient visit in 2010 Intl\$ according to the World Health Organisation for the respective country (World Health Organization 2011) and m_c denotes the marginal effect at means of over- or underestimating health on doctor visits (i.e., the difference in doctor visits between concordance and over- or underestimation according to our estimates). The term f_c denotes the fraction of individuals in the SHARE sample who overor underestimate their health, and $p_{c,t}$ is the population older than age 50 in the respective country and year according to predictions from the Wittgenstein Centre for Demography and Global Human Capital (2018).

We project that on average across the European Union, underestimation of the population 50+ will cost Intl\$ 71 million in 2020 per country and increase to Intl\$ 81 million by 2060. Overestimation will result in negative healthcare costs of approximately Intl\$ 37 million per country in 2020 and Intl\$ 45 million in 2060. Altogether, we project a net cost of Intl\$ 34 million per country in 2020. Note that although overestimation results in negative costs, these are in the short run only. In the long run, overestimation may result in individuals skipping timely screening and preventive care and lead to worse health, resulting in higher healthcare expenditures. Longer panel data will aid in evaluating the full cost of overestimation in future work.

The costs of misperception are also projected separately for each country in our sample. Figure 5 shows the total costs of underestimation in 2020, 2040, and 2060 – it is highest in Germany, Italy, and France. When dividing the total cost of underestimation by the population size, Germany, Denmark, and The Netherlands have the highest cost. Germany has a high marginal effect at means of underestimating health on doctor visits m_c along with a relatively high fraction of individuals that underestimate their health f_c . As a result, additional outpatient visits due to underestimation are predicted to cost Germany Intl\$ 503 million in 2020 and Intl\$ 538 million in 2060. Denmark and The Netherlands have much lower f_c , but their high cost per outpatient visit d_c along with large marginal effects m_c result in high public cost of underestimating health. Countries such as Poland and Czechia have much lower



Figure 5: Projected total public cost of underestimating health for the population 50+

costs because approximately similar percentages report over- and underestimation and also much lower consultation costs. Spain has the lowest predicted cost of underestimation per capita. In total, it is Intl\$ 59 million in 2020 and Intl\$ 76 million in 2060. Table 8 shows projected misperception costs for all countries in 2020, 2040, and 2060.

7 Conclusion

We utilise rich longitudinal data from 15 European countries from SHARE to explore the effects of health (mis)perception on healthcare utilisation. We categorise misperception as arising due to overconfidence or underconfidence. Following the literature in psychology, overconfidence is measured as overestimation of one's health, whereas underconfidence is defined analogously as underestimation of one's own health. Healthcare utilisation is measured as the annual number of doctor visits. In addition, we assess the relationship between misperception and OOP expenditures incurred by those who visit the doctor. Our results based on count models and log-Gamma models suggest that individuals who underestimate their health visit the doctor more often and have higher OOP expenditures than those who assess their health correctly. By contrast, survey participants who overestimate their health visit the doctor less often and have lower OOP payments.

Country	Year	Population 50+	Cost	Cost underestimating	Cost overestimating	Balance
		in 1,000	per visit	in mio. 2010 Intl\$	in mio. 2010 Intl\$	in mio. 2010 Intl\$
Austria	2020	3,725.3	46.8	22.1	-30.9	-8.8
	2040	4,444.5	46.8	26.4	-36.9	-10.5
	2060	4,726.9	46.8	28.1	-39.2	-11.2
Belgium	2020	4,554.0	44.4	31.8	-16.3	15.6
	2040	5,416.9	44.4	37.9	-19.3	18.5
	2060	5,922.0	44.4	41.4	-21.1	20.3
Czechia	2020	4,137.6	31.3	17.2	-24.1	-6.8
	2040	5,038.8	31.3	21.0	-29.3	-8.3
	2060	4,829.6	31.3	20.1	-28.1	-8.0
Denmark	2020	2,324.7	47.0	24.0	0.1	24.2
	2040	2,569.4	47.0	26.5	0.2	26.7
	2060	2,859.2	47.0	29.5	0.2	29.7
Estonia	2020	519.2	25.7	2.50	-1.7	0.9
	2040	583.6	25.7	2.80	-1.9	1.0
	2060	568.9	25.7	2.80	-1.8	0.9
France	2020	26,121.9	40.8	99.5	-50.1	49.4
	2040	30,875.5	40.8	117.6	-59.2	58.4
	2060	34,332.8	40.8	130.8	-65.8	65.0
Germany	2020	37,597.6	44.0	503.1	-83.5	419.6
	2040	40,352.9	44.0	540.0	-89.6	450.3
	2060	40,176.2	44.0	537.6	-89.2	448.4
Italy	2020	27,580.7	38.4	152.9	-35.8	117.1
	2040	30,070.5	38.4	166.7	-39.0	127.7
	2060	28,662.7	38.4	158.9	-37.2	121.7
Luxembourg	2020	214.5	89.5	1.7	-2.6	-0.9
	2040	313.5	89.5	2.5	-3.7	-1.3
	2060	386.3	89.5	3.0	-4.6	-1.6
Netherlands	2020	7,082.6	48.1	65.0	-47.0	18.0
	2040	8,138.9	48.1	74.7	-54.0	20.7
	2060	8,750.5	48.1	80.4	-58.1	22.3
Poland	2020	14,398.8	25.5	45.5	-70.2	-24.7
	2040	17,587.8	25.5	55.5	-85.7	-30.2
	2060	16,498.9	25.5	52.1	-80.4	-28.3
Slovenia	2020	880.7	32.6	4.9	-4.4	0.6
	2040	1,048.1	32.6	5.9	-5.2	0.7
	2060	1,026.1	32.6	5.8	-5.1	0.7
Spain	2020	19,709.0	37.8	59.0	-186.8	-127.7
	2040	24,851.7	37.8	74.5	-235.5	-161.1
	2060	25,235.6	37.8	75.6	-239.2	-163.6
Sweden	2020	3914.8	45.8	21.1	-9.2	11.9
	2040	4703.5	45.8	25.4	-11.1	14.3
	2060	5419.9	45.8	29.2	-12.7	16.5
Switzerland	2020	3,513.4	55.2	17.6	8.7	26.2
	2040	4,516.4	55.2	22.6	11.1	33.7
	2060	4,986.1	55.2	24.9	12.3	37.2

Table 8: Projected total cost of health misperception by country

Our results are robust to a range of sensitivity analyses with different model specifications, sample compositions, estimation methods, and health dimensions. In addition, we account for potential endogeneity by exploiting the panel structure of our data. Specifically, arguments concerning individuals' health perception improving as a result of frequent doctor visits do not apply because we focus on current misperception and future doctor visits. Descriptive cross-tabulations show that individual characteristics such as education, illnesses, age, retirement, supplementary insurance, and others do not matter for the relationship between health misperception and healthcare utilisation; regressions controlling for these variables confirm the stability of the results.

The main limitation of this study is related to panel attrition. Individuals who suffer from diseases are less likely to participate in consecutive survey waves and thus are less likely to be included in our sample. However, we address this limitation by running our analyses separately by the number of diseases that a participant is suffering from and find no difference in the results between healthy and unhealthy participants, which suggests that panel attrition is not a concern in our study. Future work could fruitfully explore the long-term effects of health misperception on healthcare utilisation, for example, exploiting national panel data collected over a longer period of time than SHARE data. Longer panels would also allow for panel regressions and thus enable researchers to control for unobserved heterogeneity between observations.

The policy implications of our results are straightforward. First, addressing rising health expenditures has been a top priority on policymakers' agenda in many countries. Excessive hospital admissions use more than 37 million bed days across the European Union every year, significantly increasing public expenditures (OECD & European Commission 2018). Containing sources of waste and inefficiency in healthcare on either the demand or supply side is important in this regard. Our paper provides new insights, highlighting demand-side misperception as a possible source of wasteful spending. Given our results, we perform a back-of-the-envelope calculation of the costs of health misperception. (see Appendix 6 for detail). We project that on average across the European Union, underestimation will cost Intl\$ 71 million in 2020 per country and increase to Intl\$ 81 million by 2060. While overestimating health reduces public healthcare expenditure in the short run, it is likely that in the long run it will increase cost due to forgone preventive care.

Second, if individuals' own perceptions of health are what drive healthcare demand beyond actual health and other socioeconomic characteristics, then equipping them through person-

alised or public health campaigns with the necessary tools and information to accurately assess their health and determine when to seek healthcare is perhaps a valuable long-term strategy for reducing unnecessary healthcare use. This is a particularly relevant measure in countries with ageing populations that suffer from cognitive dissonance and thereby increased health misperception (Brandtstädter & Greve 1994, Frieswijk et al. 2004, Henchoz et al. 2008, Idler 1993, Spitzer & Weber 2019). Reaching out to those who overestimate their health by providing information about the benefits of screening and preventive care might also improve their health and thus prevent suffering and costs in the long run. Initiatives to increase health literacy, such as the National Action Plan on Health Literacy, are already in place in Germany (Vogt et al. 2018). Other countries can follow similar approaches to evaluate health literacy levels and take strategic action to educate people.

Finally, wait time is often used as a non-price rationing measure in healthcare by policymakers (Barzel 1974, Iversen & Siciliani 2011). Identifying patients with health misperception and reducing unnecessary visits to the doctor can have important implications for the effectiveness of such rationing mechanisms. Not only will they free up physician capacity, but they can also directly ensure timely care for other patients who are in need of urgent intervention. Moreover, with the advent of artificial intelligence and technology, providing individuals with the option to use online physician chatbots and telephone consultations will further reduce the burden of unnecessary doctor visits due to misperception rather than true health need.

References

- Anderson, A., Baker, F. & Robinson, D. T. (2017), 'Precautionary savings, retirement planning and misperceptions of financial literacy', *Journal of Financial Economics* 126(2), 383– 398.
- Arni, P., Dragone, D., Goette, L. & Ziebarth, N. R. (2019), 'Biased Health Perceptions and Risky Health Behaviors — Theory and Evidence', *Preliminary Draft*.
- Avitabile, C., Jappelli, T. & Padula, M. (2011), 'Cognitive abilities, healthcare and screening tests', *Journal of Population Ageing* **4**, 251–269.
- Bago d'Uva, T. & Jones, A. M. (2009), 'Health care utilisation in Europe: New evidence from the ECHP', *Journal of Health Economics* **28**(2), 265–279.
- Bago d'Uva, T., O'Donnell, O. & Van Doorslaer, E. (2008), 'Differential health reporting by education level and its impact on the measurement of health inequalities among older Europeans', *International Journal of Epidemiology* **37**(6), 1375–1383.
- Barzel, Y. (1974), 'A Theory of Rationing by Waitin', *The Journal of Law & Economics* **17**(1), 73–95.
- Baumann, A. O., Deber, R. B. & Thompson, G. G. (1991), 'Overconfidence among physicians and nurses: The 'micro-certainty, macro-uncertainty' phenomenon', *Social Science and Medicine* 32(2), 167–174.
- Beaudoin, M. & Desrichard, O. (2011), 'Are memory self-efficacy and memory performance related? A meta-analysis.', *Psychological Bulletin* 137(2), 211–241.
- Berner, E. S. & Graber, M. L. (2008), 'Overconfidence as a Cause of Diagnostic Error in Medicine', American Journal of Medicine 121(5 SUPPL.), S2–S23.
- Bíró, A. (2013), 'Discount rates and the education gradient in mammography in the UK', *Health economics* (22), 1021–1036.
- Bíró, A. (2014), 'Supplementary private health insurance and health care utilization of people aged 50+', *Empirical Economics* **46**(2), 501–524.
- Bíró, A. (2016*a*), 'Differences between subjective and predicted survival probabilities and their relation to preventive care use', *B.E. Journal of Economic Analysis and Policy* **16**(2), 807– 835.
- Bíró, A. (2016*b*), 'Outpatient visits after retirement in Europe and the US', *International Journal of Health Economics and Management* **16**(4), 363–385.

- Bolin, K., Lindgren, A., Lindgren, B. & Lundborg, P. (2009), 'Utilisation of physician services in the 50+ population: The relative importance of individual versus institutional factors in 10 European countries', *International Journal of Health Care Finance and Economics* 9(1), 83– 112.
- Börsch-Supan, A. (2019*a*), 'Survey of Health, Ageing and Retirement in Europe (SHARE)Wave 1. Release version: 7.0.0. SHARE-ERIC. Data set.'.URL: 10.6103/SHARE.w1.700
- Börsch-Supan, A. (2019b), 'Survey of Health, Ageing and Retirement in Europe (SHARE)Wave 4. Release version: 7.0.0. SHARE-ERIC. Data set.'.URL: 10.6103/SHARE.w4.700
- Börsch-Supan, A. (2019c), 'Survey of Health, Ageing and Retirement in Europe (SHARE)Wave 6. Release version: 7.0.0. SHARE-ERIC. Data set.'.URL: 10.6103/SHARE.w6.700
- Börsch-Supan, A., Brandt, M., Hunkler, C., Kneip, T., Korbmacher, J., Malter, F., Schaan, B., Stuck, S. & Zuber, S. (2013), 'Data resource profile: The Survey of Health, Ageing and Retirement in Europe (SHARE)', *International Journal of Epidemiology* 42(4), 992–1001.
- Brandtstädter, J. & Greve, W. (1994), 'The aging self: Stabilizing and protecitve processes', *Developmental Review* (14), 52–80.
- Buchmueller, T. C., Couffinhal, A., Grignon, M. & Perronnin, M. (2004), 'Access to physician services: Does supplemental insurance matter? Evidence from France', *Health Economics* 13, 669–687.
- Cameron, K. A., Song, J., Manheim, L. M. & Dunlop, D. D. (2010), 'Gender disparities in health and healthcare use among older adults', *Journal of Women's Health* **19**(9), 1643–1650.
- Capistrant, B. D., Glymour, M. M. & Berkman, L. F. (2014), 'Assessing mobility difficulties for cross-national comparisons: Results from the WHO study on AGEing and Adult Health', *Journal of the American Geriatrics Society* 62(2), 329–335.
- Choi, A. & Cawley, J. (2017), 'Health disparities across education: The role of differential reporting error', *Health Economics* **27**(3), e1–e29.
- Christensen, K., Doblhammer, G., Rau, R. & Vaupel, J. W. (2009), 'Ageing populations: the challenges ahead', *Lancet* **374**(9696), 1196–1208.
- Coman, L. & Richardson, J. (2006), 'Relationship between self-report and performance measures of function: a systematic review.', *Canadian journal on aging* **25**(3), 253–270.

- Coulson, E., Terza, J. V., Neslusan, C. A. & Stuart, B. C. (1995), 'Estimating the Moral-Hazard Effect of Supplemental Medical Insurance in the Demand for Prescription Drugs by the Elderly', *American Economic Review* **85**, 122–126.
- Cox, N. (2011), 'Stata tip 96: Cube roots', Stata Journal 11(1), 149–154.
- Crossley, T. F. & Kennedy, S. (2001), 'The reliability of self-assessed health status.', *Journal of Health Economics* **21**, 643–658.
- Cutler, D., Skinner, J., Stern, A. D. & Wennberg, D. (2013), 'Physician beliefs and patient preferences : A new look at regional variation in health care spending', *NBER Working Paper Series* (19320).
- Deb, P. & Norton, E. C. (2018), 'Modeling Health Care Expenditures and Use', *Annual Review* of *Public Health Modeling* **39**, 489–505.
- European Commission (2018), The 2018 Ageing Report: Economic & budgetary projections for the 28 EU member states (2016-2070), Technical report.
- Eurostat (2018), 'International Standard Classification of Education (ISCED)'.
 URL: http://ec.europa.eu/eurostat/statistics-explained/index.php/International_Standard
 _Classification_of_Education_%28ISCED%29#Correspondence_between_ISCED_2011
 _and_ISCED_1997
- Eurostat (2019), *Ageing Europe: Looking at the lives of older People in the EU*, Publications Office of the European Union, Luxembourg.
- Ferrer, M., Lamarca, R., Orfila, F. & Alonso, J. (1999), 'Comparison of performance-based and self-rated functional capacity in Spanish elderly', *American Journal of Epidemiology* 149(3), 228–235.
- Frieswijk, N., Buunk, B. P., Steverink, N. & Slaets, J. P. (2004), 'The effect of social comparison information on the life satisfaction of frail older persons', *Psychology and Aging* 19(1), 183– 190.
- Furnham, A. (2001), 'Self-estimates of intelligence : culture and gender difference in self and other estimates of both general (g) and multiple intelligences', *Personality and Individual Differences* **31**, 1381–1405.
- Galdas, P. M., Cheater, F. & Marshall, P. (2005), 'Men and health help-seeking behaviour: Literature review', *Journal of Advanced Nursing* **49**(6), 616–623.

- Gardner, R. C., Langa, K. M. & Yaffe, K. (2017), 'Subjective and objective cognitive function among older adults with a history of traumatic brain injury: A population-based cohort study', *PLoS Medicine* **14**(3), 1–16.
- Grodstein, F., Chen, J., Wilson, R. S. & Manson, J. E. (2001), 'Type 2 diabetes and cognitive function in community-dwelling elderly women', *Diabetes Care* **24**(6), 1060–1065.
- Henchoz, K., Cavalli, S. & Girardin, M. (2008), 'Health perception and health status in advanced old age: A paradox of association', *Journal of Aging Studies* **22**(3), 282–290.
- Idler, E. L. (1993), 'Age differences in self-assessments of health: Age changes, cohort differences, or survivorship?', *Journal of Gerontology* **48**(6), 289–300.
- Iversen, T. & Siciliani, L. (2011), Non-Price Rationing and Waiting Times, *in* S. Glied & P. C. Smith, eds, 'The Oxford Handbook of Health Economics', Oxford University Press, Oxford New York.
- Jackson, J. D., Rentz, D. M., Aghjayan, S. L., Buckley, R. F., Meneide, T. F., Sperling, R. A. & Amariglio, R. E. (2017), 'Subjective cognitive concerns are associated with objective memory performance in Caucasian but not African-American persons', *Age and Ageing* 46(6), 988–993.
- Johnson, D. D. P. & Fowler, J. H. (2011), 'The evolution of overconfidence', *Nature* **477**, 317–320.
- Jürges, H. (2007), 'True health vs response style: exploring cross-country differences in self-reported health', *Health economics* **16**, 163–178.
- Jürges, H. (2015), Health care utilization and out-of-pocket expenses, *in* F. Malter & A. Börsch-Supan, eds, 'SHARE Wave 5 : Innovations & Methodology', Munich Center for the Economics of Ageing (MEA) at the Max Planck Institute for Social Law and Socia, Munich, pp. 37–42.
- Lange, F. (2011), 'The role of education in complex health decisions: Evidence from cancer screening', *Journal of Health Economics* **30**(1), 43–54.
- Lugo-Palacios, D. G. & Gannon, B. (2017), 'Health care utilisation amongst older adults with sensory and cognitive impairments in Europe', *Health Economics Review* 7(1), 1–15.
- Lutz, W., O'Neill, B. C. & Scherbov, S. (2003), 'Europe's population at a turning point', *Science* **299**, 1991–1992.
- Mansfield, A., Addis, M. & Mahalik, J. (2003), "Why Won't He Go to the Doctor?": The Psychology of Men's Help Seeking', *International Journal of Men's Health* **2**(2), 93–109.

- Merrill, S. S., Seeman, T. E., Kasl, S. V. & Berkman, L. F. (1997), 'Gender differences in the comparison of self-reported disability and performance measures', *Journals of Gerontology Series A Biological Sciences and Medical Sciences* 52(1), 19–26.
- Meyer, A. N., Payne, V. L., Meeks, D. W., Rao, R. & Singh, H. (2013), 'Physicians' diagnostic accuracy, confidence, and resource requests: A vignette study', *JAMA Internal Medicine* 173(21), 1952–1961.
- Moore, D. A. & Healy, P. J. (2008), 'The Trouble With Overconfidence', *Psychological Review* **115**(2), 502–517.
- Moreira, S. & Barros Pita, P. (2010), 'Double health insurance coverage and health care utilisation: Evidence from quantile regression', *Health economics* **19**, 1075–1092.
- OECD & European Commission (2018), *Health at a Glance: Europe 2018: State of health in the EU cycle*, OECD Publishing, Paris.
- Ortoleva, P. & Snowberg, E. (2015), 'Overconfidence in political behavior', *American Economic Review* **105**(2), 504–535.
- Paccagnella, O., Rebba, V. & Weber, G. (2013), 'Voluntary private health insurance among the over 50s in Europe', *Health economics* **22**, 289–315.
- Picone, G., Sloan, F. & Taylor, D. (2004), 'Effects of risk and time preference and expected longevity on demand for medical tests', *Journal of Risk and Uncertainty* **28**(1), 39–53.
- Pinheiro, P. A., Carneiro, J. A., Coqueiro, R. S., Pereira, R. & Fernandes, M. H. (2016), 'Chair stand test as simple tool for sarcopenia screening in elderly women', *Journal of Nutrition*, *Health and Aging* 20(1), 56–59.
- Preston, C. E. & Harris, S. (1965), 'Psychology of drivers in traffic accidents', *Journal of Applied Psychology* **49**(4), 284–288.
- Purser, J. L., Fillenbaum, G. G., Pieper, C. F. & Wallace, R. B. (2005), 'Mild cognitive impairment and 10-year trajectories of disability in the Iowa established populations for epidemiologic studies of the elderly cohort', *Journal of the American Geriatrics Society* 53(11), 1966–1972.
- Reuben, E., Wiswall, M. & Zafar, B. (2017), 'Preferences and Biases in Educational Choices and Labour Market Expectations: Shrinking the Black Box of Gender', *Economic Journal* 127(604), 2153–2186.

- Rocca, P., Beckman, A., Ekvall Hansson, E. & Ohlsson, H. (2015), 'Is the association between physical activity and healthcare utilization affected by self-rated health and socioeconomic factors?', *BMC Public Health* **15**(1), 1–8.
- Sainio, P., Koskinen, S., Heliövaara, M., Martelin, T., Härkänen, T., Hurri, H., Miilunpalo, S. & Aromaa, A. (2006), 'Self-reported and test-based mobility limitations in a representative sample of Finns aged 30+', *Scandinavian Journal of Public Health* 34(4), 378–386.
- Sakurai, R., Fujiwara, Y., Ishihara, M., Higuchi, T., Uchida, H. & Imanaka, K. (2013), 'Agerelated self-overestimation of step-over ability in healthy older adults and its relationship to fall risk', *BMC Geriatrics* **13**(44), 1–9.
- Schlichthorst, M., Sanci, L. A., Pirkis, J., Spittal, M. J. & Hocking, J. S. (2016), 'Why do men go to the doctor? Socio-demographic and lifestyle factors associated with healthcare utilisation among a cohort of Australian men', *BMC Public Health* **16**(Suppl 3), 81–90.
- Schneider, U., Pfarr, C., Schneider, B. S. & Ulrich, V. (2012), 'I feel good! Gender differences and reporting heterogeneity in self-assessed health', *European Journal of Health Economics* 13(3), 251–265.
- Spitzer, S. & Weber, D. (2019), 'Reporting biases in self-assessed physical and cognitive health status of older Europeans', *Plos One* **14**(10), e0223526.
- Srisurapanont, M., Suttajit, S., Eurviriyanukul, K. & Varnado, P. (2017), 'Discrepancy between objective and subjective cognition in adults with major depressive disorder', *Scientific Reports* 7(1), 1–7.
- Steel, N., Huppert, F. & Melzer, D. (2003), Physical and cognitive function, *in* M. Marmot, J. Banks, R. Blundell, C. Lessof & J. Nazroo, eds, 'Health, Wealth and Lifestyles of the Older Population in England: The 2002 English Longitudinal Study of Ageing', IFS, London, chapter Physical a.
- Tavares, L. P. & Zantomio, F. (2017), 'Inequity in healthcare use among older people after 2008: The case of southern European countries', *Health Policy* **121**(10), 1063–1071.
- Vallejo-Torres, L. & Morris, S. (2013), 'Income-related inequity in healthcare utilisation among individuals with cardiovascular disease in England - account for vertical inquity', *Health economics* (22), 533–553.
- Van Doorslaer, E., Koolman, X. & Jones, A. M. (2004), 'Explaining income-related inequalities in doctor utilisation in Europe', *Health Economics* **13**(7), 629–647.

- Vogt, D., Schaeffer, D., Bauer, U., Koplatzik, K. & Hurrelmann, K. (2018), 'The German National Action Plan on health literacy: Empirical base, development and implementation', *European Journal of Public Health* **28**.
- Wittgenstein Centre for Demography and Global Human Capital (2018), 'Wittgenstein Centre Data Explorer Version 2.0 (Beta)'.URL: http://www.wittgensteincentre.org/dataexplorer
- World Health Organization (2011), 'Country-specific inpatient and outpatient estimates in
 2010 currency'.
 URL: https://www.who.int/choice/cost-effectiveness/inputs/ country_inpatient_outpatient_2010.pdf?ua=1
- Zell, E. & Krizan, Z. (2014), 'Do People Have Insight Into Their Abilities? A Metasynthesis', *Perspectives on Psychological Science* 9, 111–125.
- Zhang, Y., Salm, M. & Soest, A. V. (2018), 'The effect of retirement on healthcare utilization : Evidence from China', *Journal of Health Economics* **62**, 165–177.

A Appendix

		Heal	th perception		
	Pos. concordance	Underestimating Mear	Neg. concordance doctor visits	Overestimating	Total
No. chronic diseases at w					
0	4.6	7.2	10.5	6.1	5.1
1	6.3	9.5	10.5	8.0	7.0
2	8.1	10.6	12.1	11.0	9.0
3	8.8	11.3	11.7	12.4	10.0
4	11.0	16.2	13.8	15.5	13.1
5	13.5	13.5	16.2	11.3	13.9
6	11.7	18.9	14.8	17.2	14.6
7	11.0	14.1	26.9	32.1	21.1
8	11.0	19.5	15.0	14.9	13.2
9	20.0	19.0	10.0	14.7	20.0
10	20.0				11 7
Total	6.2	10.1	11.9	8.6	7.2
No activity limitations at w					
	60	9.0	11.1	78	65
1	83	10.3	10.8	9.8	0.0
1	0.5	10.5	10.0	12.0	12.0
2	9.0 11 7	14.0	11.0	15.9	12.0
4	11.7	21.1	11.9	10.1	12.2
4	10.2	51.1	11.0	12.5	13.5
5	7.8	11./	12.3	18.6	12.8
6	5.2	11.4	14./	12.4	13.2
7	8.8	15.8	13.4	15.6	13.8
8	4./	8.6	15.3	9.8	13.6
9	7.6	9.6	20.2	11.5	18.9
10		30.0	20.5	6.7	19.4
11		9.1	9.8	6.0	9.7
12		6.2	16.6	30.6	16.4
13	5.5	8.6	14.5	98.0	10.3
Total	6.2	10.1	11.8	8.6	7.2
5-year age groups					
50-54	5.2	10.6	11.7	6.8	5.9
55-59	5.4	10.9	13.8	7.2	6.4
60-64	6.0	9.1	10.5	8.9	6.8
65-69	6.8	9.5	13.5	9.4	7.7
70-74	7.7	10.8	10.9	9.1	8.5
75-79	8.1	9.0	12.0	12.1	9.1
80-84	8.0	11.4	10.4	9.0	9.0
85-89	6.4	9.4	12.8	9.3	8.7
90+	5.5	7.8	9.2	13.9	9.0
Total	6.2	10.1	11.8	8.6	7.2
Gender					
Men	6.0	10.0	11.8	8.1	6.7
Women	6.4	10.1	11.9	9.1	7.5
Total	6.2	10.1	11.8	8.6	7.2
Education					
Low	6.6	10.1	11.8	9.2	7.7
Medium	6.2	9.7	12.1	8.2	7.0
High	5.7	10.8	11.3	7.5	6.4
Total	6.2	10.1	11.8	8.6	7.2

Table A.1: Crosstable mean doctor visits at w + 1 (weighted)

		Health perception						
	Pos. concordance	Underestimating Mea	Neg. concordance n doctor visits	Overestimating	Total			
Is retired								
No	5.5	10.2	11.6	7.9	6.4			
Yes	7.0	10.0	11.9	9.3	7.9			
Total	6.2	10.1	11.8	8.6	7.2			
Is married								
No	6.4	9.9	11.5	9.2	7.5			
Yes	6.1	10.2	12.0	8.5	7.0			
Total	6.2	10.1	11.8	8.7	7.2			
Healt access								
Not difficult	5.9	10.2	11.0	8.4	6.7			
Difficult	6.5	10.0	12.0	10.6	8.3			
Total	6.0	10.1	11.3	8.9	7.0			
Supplementary insurance								
No	6.1	10.4	11.7	9.7	7.3			
Yes	5.7	9.4	10.6	7.1	6.4			
Total	6.0	10.0	11.4	8.9	7.0			
Has children								
No	6.1	10.8	12.0	10.1	7.3			
Yes	6.2	10.0	11.8	8.5	7.2			
Total	6.2	10.1	11.8	8.7	7.2			
Country								
Austria	6.1	8.8	11.4	8.9	7.0			
Germany	6.9	11.2	13.4	10.1	8.0			
Sweden	3.8	5.7	10.6	5.4	4.3			
Netherlands	5.0	7.4	9.8	8.6	5.5			
Spain	5.2	8.6	10.2	7.5	6.2			
Italy	7.3	13.1	13.8	10.2	8.6			
France	5.6	7.7	9.3	6.7	6.1			
Denmark	4.3	8.2	10.6	6.4	4.9			
Switzerland	4.4	7.2	8.6	8.0	5.0			
Belgium	7.0	10.3	16.4	9.5	8.0			
Czechia	6.5	9.7	11.2	9.1	7.5			
Poland	6.7	9.9	10.2	6.8	7.5			
Luxembourg	7.7	10.5	17.3	12.1	9.0			
Slovenia	4.5	7.4	9.0	7.5	5.5			
Estonia	4.9	7.6	8.4	6.2	5.9			
Total	6.2	10.1	11.8	8.6	7.2			
Survey wave								
Wave 2	6.5	10.1	12.6	8.3	7.4			
Wave 5	6.0	10.0	11.4	8.9	7.0			
Total	6.2	10.1	11.8	8.6	7.2			

Table A.1, continued: Crosstable mean doctor visits at w + 1 (weighted)

	Health perception								
	Pos. concordance Mean OOP	Underestimating Mean OOP	Neg. concordance Mean OOP	Overestimating Mean OOP	Total Mean OOP				
No. chronic diseases at w									
0	64.1	98.9	65.0	73.5	66.7				
1	73.9	100.4	56.3	66.5	75.2				
2	75.2	79.6	65.8	79.0	75.3				
3	80.4	76.5	94.4	102.0	83.6				
4	79.4	76.9	58.8	83.3	74.8				
5	81.1	158.1	95.0	41.3	98.1				
6	71.7	181.6	352.4	6.4	156.3				
7	38.3	95.1	9.0	67.7	40.2				
8	23.2	334.6	50.7	8.8	40.6				
10	0.0				0.0				
Total	70.8	92.4	74.6	74.6	73.6				
No. activity limitations at w									
0	71.8	90.5	75.7	73.7	73.5				
1	62.5	96.5	72.9	61.7	71.3				
2	53.6	50.9	52.3	108.2	60.7				
3	44.8	140.5	160.1	103.2	121.3				
4	35.6	86.4	34.0	34.6	46.9				
5	38.1	57.9	75.4	57.3	64.6				
6	9.8	54.0	90.8	42.6	73.3				
7	0.0	123.1	18.7	527.8	60.6				
8	34.2	0.0	47.2	59.3	39.9				
9	163.4	19.0	15.6	0.0	16.7				
10		0.0	219.9	0.0	197.9				
11		352.6	28.2	0.0	60.5				
12		289.9	46.6	8.6	100.2				
13	2.2	298.1	141.1		240.2				
Total	70.8	92.4	74.5	74.4	73.5				
5-year age groups									
50-54	60.8	147.3	57.4	46.8	65.8				
55-59	67.4	80.9	72.5	103.9	71.0				
60-64	70.6	78.8	86.0	68.4	71.8				
65-69	73.9	80.6	110.8	89.8	77.6				
70-74	89.7	99.7	86.3	58.6	88.1				
75-79	76.2	96.6	102.2	81.5	82.7				
80-84	67.5	74.3	42.5	86.9	67.1				
85-89	48.3	70.0	45.3	52.9	51.5				
90+	49.0	15.4	10.8	47.2	35.1				
Total	70.8	92.4	74.5	74.4	73.5				
Gender									
Men	74.5	70.4	62.2	60.9	72.7				
Women	67.1	104.2	79.6	84.6	74.2				
Total	70.8	92.4	74.5	74.4	73.5				
Education									
Low	62.0	89.3	78.5	64.3	66.8				
Medium	71.1	92.0	71.0	70.6	73.3				
High	85.4	108.5	55.7	120.8	88.4				
Total	71.1	93.7	74.3	74.1	73.9				

Table A.2: Crosstable mean OOP expenditures in Euros at w + 1 (weighted)

	Health perception									
	Pos. concordance Mean OOP	Underestimating Mean OOP	Neg. concordance Mean OOP	Overestimating Mean OOP	Total Mean OOP					
Is retired										
No	64.8	101.1	54.4	81.8	68.5					
Yes	77.0	87.4	88.7	70.0	78.5					
Total	70.7	92.8	75.4	75.0	73.6					
Is married										
No	69.5	72.9	42.0	68.5	67.2					
Yes	72.5	100.8	103.1	81.5	77.5					
Total	71.6	90.8	74.8	76.2	74.1					
Healt access										
Not difficult	71.5	98.6	78.2	84.0	75.4					
Difficult	67.7	81.4	66.7	43.8	66.6					
Total	71.0	95.0	74.1	74.5	73.9					
Supplementary insurance										
No	70.5	101.1	73.0	75.8	74.3					
Yes	71.2	76.6	80.2	71.3	72.2					
Total	70.7	92.5	74.6	74.5	73.5					
Has children										
No	89.3	131.7	92.9	73.9	91.6					
Yes	68.3	88.9	72.7	74.8	71.3					
Total	70.8	92.8	75.0	74.7	73.6					
Country										
Austria	123.9	141.1	124.0	58.3	121.0					
Germany	55.8	45.6	30.7	75.9	54.2					
Sweden	67.9	75.2	75.6	66.9	68.9					
Spain	10.1	33.4	64.2	16.5	18.0					
Italy	139.7	270.5	135.8	132.5	149.2					
France	28.6	37.3	27.2	29.8	29.5					
Denmark	5.5	2.9	4.9	1.5	5.1					
Switzerland	386.5	442.0	636.0	380.6	397.0					
Belgium	90.6	183.0	155.2	90.9	105.1					
Czechia	7.3	11.2	9.6	10.9	8.3					
Luxembourg	171.3	209.6	292.1	206.8	185.3					
Slovenia	13.9	20.2	9.9	2.8	13.2					
Estonia	14.8	12.0	14.6	13.9	14.3					
Total	70.8	92.4	74.5	74.4	73.5					
Survey wave										
Wave 5	70.8	92.4	74.5	74.4	73.5					
Total	70.8	92.4	74.5	74.4	73.5					

Table A.2, continued: Crosstable mean OOP expenditures in Euros at w + 1 (weighted)

Ta	able A.3: Ci	rosstable	e health	percepti	on (weig	shted)	
				Heal	th percep	tion	

	-								
	Pos. con	cordance	Undere	estimating	Neg. co	ncordance	Overe	stimating	Total
	KOW %	95% CI	KOW %	95% CI	KOW %	95% CI	KOW %	95% CI	KOW %
Objective impairment									
Unimpaired (n=47,913)	88.0	[87.5,88.5]	12.0	[11.5,12.5]	0.0		0.0		100.0
Impaired (n=8,239)	0.0		0.0		38.3	[36.6,40.0]	61.7	[60.0,63.4]	100.0
Total (n=56,152)	74.3	[73.6,74.9]	10.1	[9.7,10.5]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Bearson: Uncompated abi2(2) -	E 62a + 04								
Pearson: Uncorrected $\operatorname{Chi}_{2}(5) =$	5.02e+04	D	0.000						
Design-based $F(2.77, 155429.77) =$	/904.22//	rr=	0.000						
No. chronic diseases at w									
0 (n=20,630)	82.1	[81.2,82.9]	5.7	[5.2,6.2]	3.1	[2.7,3.5]	9.1	[8.5 <i>,</i> 9.9]	100.0
1 (n=17,715)	76.2	[75.1,77.3]	10.2	[9.5,11.0]	4.4	[3.9,4.9]	9.2	[8.5,10.0]	100.0
2 (n=10,246)	67.8	[66.2,69.3]	13.9	[12.8,15.1]	8.2	[7.4,9.1]	10.1	[9.1,11.2]	100.0
3 (n=4,712)	60.4	[58.0,62.7]	16.2	[14.6,18.0]	11.7	[10.3,13.4]	11.6	[10.1,13.3]	100.0
4 (n=1,889)	48.2	[44.4,51.9]	20.4	[17.4,23.8]	20.1	[17.2,23.3]	11.4	[9.1,14.1]	100.0
5 (n=650)	39.5	[32.7,46.8]	25.7	[19.9,32.5]	24.1	[18.8,30.4]	10.7	[7.3,15.3]	100.0
6 (n=180)	39.9	[29.2,51.8]	16.3	[10.0,25.4]	28.5	[19.2,40.2]	15.2	[8.2,26.5]	100.0
7 (n=51)	22.1	[10.0,41.8]	19.9	[7.7,42.6]	53.0	[30.4,74.5]	5.0	[1.7,13.8]	100.0
8 (n=18)	50.9	[17.5,83.5]	3.7	[0.5,22.2]	36.6	[9.0,77.1]	8.8	[1.5,38.2]	100.0
9 (n=1)	100.0		0.0		0.0		0.0		100.0
10 (n=2)	100.0		0.0		0.0		0.0		100.0
Total (n=56,094)	74.3	[73.6,74.9]	10.1	[9.7,10.5]	6.0	[5.7 <i>,</i> 6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(30) =	3717.8201								
Design-based $F(24.19, 1.36e+06) =$	56.8893	Pr =	0.000						
No. activity limitations at w	00 ([00 0 01 0]	70		2 ([2,2,2,0]	0.0		100.0
0(n=47,381)	80.6	[80.0,81.2]	7.9	[7.5,8.3]	2.6	[2.3,2.8]	8.9	[8.5,9.4]	100.0
1(n=4,717)	50.3	[47.9,52.7]	22.7	[20.8,24.9]	12.8	[11.3,14.5]	14.2	[12.5,16.0]	100.0
2(n=1,683)	30.2	[26.6,34.0]	26.5	[23.2,30.0]	29.9	[26.3,33.8]	13.5	[11.1,16.4]	100.0
3(n=851)	21.3	[16.7,26.6]	25.7	[21.2,30.7]	38.9	[33.5,44.6]	14.2	[10.4,19.0]	100.0
4(n=485)	15.4	[10.4,22.2]	20.9	[14.4,29.3]	48.6	[40.4,56.8]	15.2	[10.9,20.7]	100.0
5(n=340)	11.4	[6.9,18.2]	19.3	[13.3,27.2]	52.0	[43.0,60.9]	17.3	[11.0,26.1]	100.0
6(n=224)	7.0	[2.5,18.1]	13.9	[8.4,22.1]	63.1	[51.7,73.2]	16.0	[9.3,26.1]	100.0
2 (n=159)	4.6	[1.6,12.6]	20.0	[10.5,34.7]	69.9 75.0	[55.8,81.1]	5.4 4 E	[2.2,12.9]	100.0
8(n=93)	3.5	[0.8,13.2]	17.0	[7.8,33.0]	/5.0	[59.6,85.9]	4.5	[1.8,10.8]	100.0
9(n=/1)	0.7	[0.2,3.4]	9.3	[3.0,25.7]	87.6	[71.8,95.1]	2.4	[0.4,12.7]	100.0
10(n=42)	0.0		0.5	[0.9,34.1]	81.1 01.([38.6,92.9]	12.4	[4.0,32.4]	100.0
11(n=27)	0.0		7.4	[1.7,26.5]	91.6	[73.2,97.7]	1.1	[0.1,7.6]	100.0
12(n=26)	0.0		22.8	[8.5,48.3]	01.8	[36.9,81.7]	15.4	[4.2,43.0]	100.0
13(n=49)	4.4	[0.9, 19.7]	10.1	[49.3,79.8]	29.2	[10.8,45.7]	0.1	[0.0,0.8]	100.0
10tal (n=56,148)	74.3	[73.6,74.9]	10.1	[9.7,10.5]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(39) =	1.49e+04								
Design-based F(33.61, 1.89e+06) =	177.1679	Pr =	0.000						
5-year age groups									
50-54 (n=7.593)	81.7	[80.1.83.2]	7.0	[6.1.8.0]	2.8	[2.3.3.5]	8.5	[7.4.9.7]	100.0
55-59 (n=10.672)	78.6	[77.3.80.0]	8.7	[7.9.9.7]	4.0	[3.4.4.7]	8.6	[7.7.9.6]	100.0
60-64 (n=11,137)	77.0	[75 7 78 2]	10.6	[97116]	4 2	[3748]	82	[7490]	100.0
65-69 (n=10.290)	73.6	[72.2.74.9]	10.7	[9.8.11.6]	5.4	[4.8.6.2]	10.3	[9.4.11.3]	100.0
70-74 (n=8.143)	68.7	[67.0.70.3]	12.5	[11.3.13.8]	8.3	[7.4.9.3]	10.5	[9.5.11.7]	100.0
75-79 (n=4.390)	63.7	[61.2.66.1]	13.9	[12.3.15.8]	11.8	[10.2.13.5]	10.6	[9.2.12.3]	100.0
80-84 (n=2.645)	55.7	[52.4.59.0]	13.2	[11.1.15.6]	16.5	[14.1.19.2]	14.6	[12.3.17.2]	100.0
85-89 (n=1.047)	47.4	[42.3 52 5]	13.8	[10.7 17 7]	20.6	[16.8 25 1]	18.1	[14.6 22 2]	100.0
90+(n=2.35)	38.2	[28.3 49 1]	7.3	[4,1 12 6]	20.0	[18.9.38.2]	27.0	[18.4.37.8]	100.0
Total (n=56,152)	74.3	[73.6.74.9]	10.1	[9.7.10.5]	6.0	[5,7.6.3]	9.6	[9.2.10.1]	100.0
		[,,,]	10.1	[,,10.0]	0.0	[0.0,0.0]	2.0	[,.=,10,1]	
Pearson: Uncorrected $chi2(24) =$	2476.8374		0.000						
Design-based $F(22.74, 1.28e+06) =$	37.9231	Pr =	0.000						

				Healtl	h percep	tion			
	Pos. con	cordance	Undere	estimating	Neg. co	ncordance	Overe	stimating	Total
	Row %	95% CI	Row %	95% CI	Row %	95% CI	Row %	95% CI	Row %
Gender									
Men (n=24,503)	79.4	[78.5,80.2]	7.3	[6.8,7.8]	3.8	[3.5,4.2]	9.5	[8.9,10.2]	100.0
Women (n=31,649)	69.9	[69.0,70.7]	12.5	[11.9,13.2]	7.8	[7.3,8.3]	9.8	[9.2,10.4]	100.0
Total (n=56,152)	74.3	[73.6,74.9]	10.1	[9.7,10.5]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(3) =	922.0913								
Design-based F(2.99, 167664.12) =	107.2160	Pr =	0.000						
Education									
Low (n=21,346)	68.2	[67.2,69.2]	11.0	[10.4,11.7]	8.8	[8.3,9.5]	12.0	[11.3,12.7]	100.0
Medium (n=21,228)	76.1	[75.0,77.1]	10.3	[9.6,11.0]	4.8	[4.3,5.3]	8.8	[8.1,9.6]	100.0
High (n=12,833)	83.2	[82.0,84.3]	7.8	[7.1,8.6]	2.5	[2.1,3.0]	6.5	[5.7,7.3]	100.0
Total (n=55,407)	74.3	[73.7,74.9]	10.1	[9.7,10.5]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(6) =	1177.9308								
Design-based F(5.94, 329109.56) =	68.3745	Pr =	0.000						
Is retired									
No (n=25,298)	77.6	[76.7,78.5]	8.7	[8.2,9.3]	4.8	[4.4,5.3]	8.8	[8.2,9.5]	100.0
Yes (n=30,601)	71.0	[70.1,71.8]	11.6	[11.0,12.2]	7.0	[6.6,7.5]	10.5	[9.9,11.1]	100.0
Total (n=55,899)	74.4	[73.8,75.0]	10.1	[9.7,10.5]	5.9	[5.6,6.2]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(3) =	344.7396								
Design-based F(2.99, 167064.47) =	39.5269	Pr =	0.000						
Is married									
No (n=14,874)	69.8	[68.5,71.1]	11.4	[10.6,12.4]	8.1	[7.4,8.9]	10.7	[9.8,11.6]	100.0
Yes (n=39,474)	76.1	[75.4,76.7]	9.6	[9.2,10.1]	5.1	[4.8,5.5]	9.2	[8.8,9.7]	100.0
Total (n=54,348)	74.2	[73.5,74.8]	10.2	[9.7,10.6]	6.0	[5.7 <i>,</i> 6.3]	9.7	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(3) =	290.4041								
Design-based F(2.98, 162037.55) =	30.4726	Pr =	0.000						
Healt access									
Not difficult (n=32,586)	78.2	[77.4,78.9]	9.7	[9.2,10.3]	5.0	[4.6,5.4]	7.2	[6.7,7.7]	100.0
Difficult (n=6,341)	60.8	[58.5,63.0]	13.1	[11.7,14.7]	14.6	[13.1,16.4]	11.5	[10.1,13.0]	100.0
Total (n=38,927)	75.3	[74.6,76.1]	10.3	[9.7,10.8]	6.5	[6.1,7.0]	7.9	[7.4,8.4]	100.0
Pearson: Uncorrected chi2(3) =	1158.8844								
Design-based F(2.99, 116441.54) =	116.8392	Pr =	0.000						

Table A.3, continued: Crosstable health perception (weighted)

				Healtl	h percep	tion			
	Pos. con	cordance	Undere Dour %	estimating	Neg. co	ncordance	Overes	stimating	Total
	KOW 70	95% CI	KOW 70	95% CI	KOW 70	95% CI	KOW 70	95% CI	KOW 70
Supplementary insurance									
No (n=26,149)	73.0	[72.1,74.0]	10.3	[9.6,10.9]	7.8	[7.3,8.4]	8.9	[8.3 <i>,</i> 9.5]	100.0
Yes (n=15,280)	78.8	[77.6,79.9]	10.4	[9.6,11.2]	4.4	[3.8,4.9]	6.5	[5.9,7.2]	100.0
Total (n=41,429)	75.1	[74.4,75.8]	10.3	[9.8,10.8]	6.6	[6.2,7.0]	8.0	[7.6,8.5]	100.0
Pearson: Uncorrected chi2(3) =	282.3894								
Design-based F(2.99, 123937.16) =	31.5801	Pr =	0.000						
Has children									
No (n=5,121)	74.2	[72.1,76.1]	9.3	[8.0,10.7]	6.0	[5.0,7.2]	10.5	[9.2,12.1]	100.0
Yes (n=50,336)	74.2	[73.6,74.9]	10.2	[9.8,10.7]	6.0	[5.7,6.3]	9.5	[9.1,10.0]	100.0
Total (n=55,457)	74.2	[73.6,74.8]	10.1	[9.7,10.6]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(3) =	10.8682								
Design-based F(2.99, 165995.64) =	1.0866	Pr =	0.353						
Country									
Austria (n=3,241)	72.7	[70.7,74.6]	11.5	[10.1,12.9]	7.1	[6.1,8.3]	8.7	[7.5,10.0]	100.0
Germany $(n=5.222)$	75.8	[74.4.77.2]	12.7	[11.6.13.8]	5.1	[4.4.5.8]	6.4	[5.7.7.2]	100.0
Sweden (n=4.722)	80.3	[78.9.81.6]	9.6	[8.6.10.5]	3.9	[3.3.4.7]	6.2	[5.4.7.1]	100.0
Netherlands $(n=1.376)$	84.2	[82.1.86.2]	8.5	[7.1.10.1]	2.9	[2.1.4.0]	4.4	[3.4.5.7]	100.0
Spain (n=5.384)	71.7	[69.8.73.5]	8.6	[7.5.9.7]	8.7	[7.7.9.8]	11.0	[9.8.12.4]	100.0
Italy $(n=4.868)$	70.3	[68.7.71.8]	8.3	[7.4.9.2]	6.9	[6.1.7.8]	14.5	[13.4.15.8]	100.0
France $(n=4.311)$	76.6	[75.1.78.1]	9.0	[8.1.10.0]	4.5	[3.8.5.2]	9.9	[8.8.11.0]	100.0
Denmark $(n=4.475)$	86.1	[85 0 87 1]	79	[7188]	2.6	[2232]	34	[2840]	100.0
Switzerland (n=3.344)	83.9	[82 5 85 2]	74	[6684]	2.6	[2133]	61	[5270]	100.0
Belgium $(n=5,599)$	77.0	[75 7 78 2]	11 5	[10.6.12.5]	5.2	[4 6 5 9]	6.4	[5.2,7.0]	100.0
C_{2}	71.6	[69.6.73.5]	10.2	[9 1 11 5]	83	[7395]	99	[85114]	100.0
Poland (n=1,222)	61.0	[09.0,75.0]	10.2	[9.1,11.3]	0.5	[7.5,9.5]	9.9 14 7	[0.3,11.4]	100.0
I = 1,222	72.4	[50.5,04.5]	14.1	[12.2, 10.5]	9.7	[4 0 8 2]	79	[12.7,17.0]	100.0
Elemente $(n=2,222)$	73.4	[70.3, 70.2]	12.5	[10.5, 14.0]	7.2	[4.9,0.2]	7.0	[0.1,9.9]	100.0
Sioverna (n=2,222)	72.0	[69.0,74.2]	10.0	[9.3,12.4]	12.0	[0.2,0.0]	9.9	[0.4,11.5]	100.0
Estonia $(n=4,006)$	04.9 74.2	[03.3,00.5]	15.8	[12.7, 15.0]	12.6	[11.5,13.7]	ð./	[7.8,9.7]	100.0
Iotal (n=56,152)	74.3	[73.6,74.9]	10.1	[9.7,10.5]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(42) =	1357.8340								
Design-based $F(20.44, 1.15e+06) =$	26.8724	Pr =	0.000						
Survey wave									
Wave 2 (n=14,623)	73.1	[72.0,74.1]	9.9	[9.2,10.6]	5.1	[4.7,5.7]	11.9	[11.2,12.7]	100.0
Wave 5 (n=41,529)	75.1	[74.3,75.8]	10.3	[9.8,10.8]	6.6	[6.2,7.0]	8.0	[7.6,8.5]	100.0
Total (n=56,152)	74.3	[73.6,74.9]	10.1	[9.7,10.5]	6.0	[5.7,6.3]	9.6	[9.2,10.1]	100.0
Pearson: Uncorrected chi2(3) =	272.0734								
Design-based F(3.00, 168272.56) =	30.3672	Pr =	0.000						

Table A.3, continued: Crosstable health perception (weighted)

	(1) Austria	(2) Belgium	(3) Czechia	(4) Denmark	(5) Estonia	(6) France	(7) Germany	(8) Italy
Health perception (ref.: concordance)								
Underestimating	0.157^{*}	0.180***	0.171***	0.510***	0.241***	0.173***	0.306***	0.211***
8	(0.063)	(0.051)	(0.045)	(0.090)	(0.072)	(0.048)	(0.053)	(0.063)
Chronic diseases	0.166***	0.171***	0.181***	0.206***	0.226***	0.177***	0.188***	0.218***
	(0.021)	(0.016)	(0.014)	(0.019)	(0.021)	(0.014)	(0.016)	(0.018)
Activity limitations	0.072*	0.163***	0.060*	0.075*	0.128***	0.045**	0.132***	0.030
, , , , , , , , , , , , , , , , , , ,	(0.032)	(0.029)	(0.028)	(0.036)	(0.026)	(0.017)	(0.036)	(0.017)
Age	0.006	-0.035	0.036	0.004	0.006	-0.048	-0.045	0.081*
0	(0.041)	(0.028)	(0.035)	(0.036)	(0.040)	(0.029)	(0.028)	(0.037)
Age squared	0.000	0.000	-0.000	-0.000	-0.000	0.000	0.000	-0.000
0 1	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Woman	0.094	0.134***	0.054	0.003	-0.061	0.069	0.058	0.161***
	(0.053)	(0.038)	(0.035)	(0.047)	(0.049)	(0.036)	(0.036)	(0.042)
Educ. group (ref.: low)		· · · ·	· · · ·	· · · ·	· · · ·	· · · ·	· · · ·	· · /
Medium	0.035	-0.051	0.012	-0.007	0.033	-0.035	0.040	-0.047
	(0.063)	(0.047)	(0.037)	(0.071)	(0.058)	(0.041)	(0.060)	(0.057)
High	0.050	-0.039	0.014	-0.016	0.073	-0.001	0.015	-0.178*
0	(0.070)	(0.045)	(0.052)	(0.073)	(0.074)	(0.052)	(0.065)	(0.084)
Retired	0.074	0.056	-0.025	0.045	0.038	0.001	0.097	-0.003
	(0.065)	(0.046)	(0.067)	(0.065)	(0.067)	(0.049)	(0.054)	(0.060)
Married	-0.053	-0.082	0.065	-0.120*	-0.042	-0.016	-0.013	0.007
	(0.053)	(0.042)	(0.038)	(0.050)	(0.050)	(0.039)	(0.044)	(0.059)
Equiv. hh income (cube root)	-0.000	-0.001	-0.002	-0.004	0.011	-0.002	-0.001	-0.002
1	(0.002)	(0.002)	(0.003)	(0.003)	(0.007)	(0.002)	(0.002)	(0.002)
Constant	0.867	2.702**	0.172	1.307	0.893	3.183**	3.161***	-1.246
	(1.392)	(0.925)	(1.161)	(1.249)	(1.334)	(0.969)	(0.908)	(1.214)
Wave dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	2,593	4,666	3,991	4,082	3,048	3,530	4,587	3,757
AIC	15,395	28,098	23,833	21,083	16,761	19,215	27,479	23,323
BIC	15,478	28,188	23,921	21,171	16,839	19,301	27,570	23,410

Table A.4: Underestimating health and annual number of doctor visits at w + 1 by country

Note: The coefficients are based on the sample that is objectively unimpaired, i.e. able to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.01

	(1) Luxembourg	(2) Netherlands	(3) Poland	(4) Slovenia	(5) Spain	(6) Sweden	(7) Switzerland
Health perception (ref.: concordance)					-		
Underestimating	0.088	0.370*	0.120	0.302***	0.172**	0.277***	0.240**
0	(0.105)	(0.174)	(0.079)	(0.079)	(0.057)	(0.066)	(0.079)
Chronic diseases	0.107***	0.162***	0.255***	0.222***	0.181***	0.123***	0.204***
	(0.031)	(0.043)	(0.026)	(0.028)	(0.018)	(0.022)	(0.025)
Activity limitations	0.266***	0.028	0.045	-0.002	0.038	0.159***	0.222**
,	(0.063)	(0.071)	(0.029)	(0.051)	(0.023)	(0.040)	(0.072)
Age	-0.098	-0.003	0.149	0.010	0.007	0.026	0.026
0	(0.073)	(0.129)	(0.103)	(0.058)	(0.034)	(0.040)	(0.046)
Age squared	0.001	0.000	-0.001	0.000	0.000	-0.000	-0.000
	(0.001)	(0.001)	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)
Woman	0.091	-0.054	0.038	-0.076	0.085	-0.087	-0.011
	(0.093)	(0.086)	(0.071)	(0.062)	(0.046)	(0.050)	(0.057)
Educ. group (ref.: low)							
Medium	0.228^{*}	-0.253*	0.035	0.009	-0.043	0.103	0.007
	(0.098)	(0.099)	(0.074)	(0.069)	(0.075)	(0.068)	(0.076)
High	-0.143	0.004	-0.045	-0.053	0.049	0.080	0.031
0	(0.113)	(0.112)	(0.105)	(0.090)	(0.079)	(0.069)	(0.094)
Retired	0.044	-0.236*	0.086	0.019	0.040	0.082	0.153*
	(0.104)	(0.102)	(0.086)	(0.088)	(0.050)	(0.090)	(0.077)
Married	-0.161	-0.003	0.131	0.054	0.075	-0.065	-0.147*
	(0.111)	(0.108)	(0.094)	(0.072)	(0.051)	(0.061)	(0.070)
Equiv. hh income (cube root)	0.006	-0.001	0.005	0.004	-0.001	-0.008	-0.003
	(0.003)	(0.005)	(0.007)	(0.005)	(0.003)	(0.005)	(0.002)
Constant	5.093*	1.545	-3.495	0.431	0.966	-0.038	0.578
	(2.391)	(4.046)	(3.186)	(1.906)	(1.160)	(1.388)	(1.575)
Wave dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	878	1,196	911	1,810	3,948	4,130	2,940
AIC	5,424	6,581	5 <i>,</i> 525	9,590	21,252	20,692	15,472
BIC	5,486	6,647	5,588	9,661	21,340	20,780	15,556

Table A.4, continued: Underestimating health and annual number of doctor visits at w + 1 by country

Note: The coefficients are based on the sample that is objectively unimpaired, i.e. able to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.01

	(1) Austria	(2) Belgium	(3) Czechia	(4) Denmark	(5) Estonia	(6) France	(7) Germany	(8) Italy
Health perception (ref.: concordance)								
Overestimating	-0.205	-0.112	-0.190**	0.006	-0.192**	-0.062	-0.068	-0.022
0	(0.122)	(0.084)	(0.060)	(0.157)	(0.074)	(0.085)	(0.106)	(0.090)
Chronic diseases	0.183***	0.142***	0.138***	0.203***	0.162***	0.058	0.123***	0.237***
	(0.040)	(0.027)	(0.022)	(0.038)	(0.025)	(0.029)	(0.034)	(0.031)
Activity limitations	0.044	0.110***	0.022	0.099**	0.022	0.073**	0.049*	0.048**
, ,	(0.024)	(0.021)	(0.016)	(0.034)	(0.021)	(0.027)	(0.022)	(0.015)
Age	0.034	0.013	0.087*	-0.185*	0.146*	-0.010	-0.110	0.065
õ	(0.074)	(0.055)	(0.042)	(0.081)	(0.060)	(0.056)	(0.073)	(0.051)
Age squared	-0.000	-0.000	-0.001*	0.001*	-0.001**	0.000	0.001	-0.000
0 1	(0.001)	(0.000)	(0.000)	(0.001)	(0.000)	(0.000)	(0.001)	(0.000)
Woman	-0.085	0.151	-0.077	0.176	-0.043	0.016	0.014	0.210**
	(0.135)	(0.087)	(0.064)	(0.154)	(0.073)	(0.083)	(0.109)	(0.073)
Educ. group (ref.: low)			. ,	. ,	. ,	. ,	. ,	· · ·
Medium	-0.185	-0.128	0.013	0.093	0.018	0.013	0.107	-0.149
	(0.121)	(0.100)	(0.060)	(0.194)	(0.082)	(0.093)	(0.154)	(0.109)
High	0.139	-0.110	0.066	-0.210	-0.091	0.083	-0.047	0.014
C	(0.171)	(0.094)	(0.120)	(0.181)	(0.104)	(0.126)	(0.175)	(0.146)
Retired	0.061	-0.017	-0.098	-0.014	-0.109	0.100	0.306*	-0.008
	(0.141)	(0.097)	(0.101)	(0.193)	(0.121)	(0.100)	(0.132)	(0.082)
Married	-0.004	0.150	0.055	0.171	0.185**	0.001	-0.014	-0.065
	(0.115)	(0.083)	(0.064)	(0.158)	(0.072)	(0.094)	(0.123)	(0.084)
Equiv. hh income (cube root)	-0.004	-0.013**	-0.004	-0.015	0.004	-0.001	0.005	-0.001
	(0.005)	(0.004)	(0.004)	(0.009)	(0.006)	(0.005)	(0.005)	(0.003)
Constant	1.124	1.677	-0.796	8.084**	-3.010	2.221	5.848*	-0.460
	(2.636)	(1.944)	(1.450)	(2.863)	(2.078)	(1.986)	(2.437)	(1.746)
Wave dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	423	591	964	236	834	554	532	950
AIC	2,857	4,062	6,325	1,471	5,144	3,355	3,734	6,501
BIC	2,913	4,124	6,393	1,520	5,205	3,416	3,793	6,569

Table A.5: Overestimating health and annual number of doctor visits at w + 1 by country

Note: The coefficients are based on the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.01

	(1) Luxembourg	(2) Netherlands	(3) Poland	(4) Slovenia	(5) Spain	(6) Sweden	(7) Switzerland
Health perception (ref.: concordance)							
Overestimating	-0.134	-0.346	-0.160	-0.207	-0.298***	-0.135	0.096
0	(0.166)	(0.271)	(0.112)	(0.113)	(0.071)	(0.166)	(0.149)
Chronic diseases	0.112*	0.274**	0.136***	0.073*	0.142***	0.107**	0.150*
	(0.055)	(0.102)	(0.035)	(0.035)	(0.024)	(0.040)	(0.072)
Activity limitations	0.122***	-0.065	0.054	-0.019	0.017	0.066*	0.118*
,	(0.032)	(0.058)	(0.034)	(0.021)	(0.018)	(0.031)	(0.055)
Age	0.373**	0.603	0.072	0.070	-0.006	-0.155	0.166
č	(0.130)	(0.345)	(0.130)	(0.076)	(0.043)	(0.115)	(0.104)
Age squared	-0.003**	-0.005	-0.001	-0.001	0.000	0.001	-0.001
	(0.001)	(0.003)	(0.001)	(0.001)	(0.000)	(0.001)	(0.001)
Woman	-0.057	-0.059	-0.138	0.221*	-0.125	-0.003	-0.001
	(0.180)	(0.242)	(0.117)	(0.094)	(0.087)	(0.154)	(0.157)
Educ. group (ref.: low)			. ,	. ,	. ,	. ,	. ,
Medium	0.164	0.476	0.005	-0.083	-0.130	0.290	-0.018
	(0.168)	(0.324)	(0.110)	(0.105)	(0.149)	(0.217)	(0.192)
High	-0.428	-0.294	0.238	-0.409*	-0.179	-0.197	-0.146
C .	(0.409)	(0.274)	(0.240)	(0.184)	(0.184)	(0.161)	(0.227)
Retired	-0.630**	-0.329	0.183	0.134	0.049	0.496*	0.122
	(0.214)	(0.331)	(0.122)	(0.121)	(0.079)	(0.238)	(0.225)
Married	-0.083	-0.179	0.044	-0.044	0.056	-0.032	-0.302
	(0.173)	(0.274)	(0.118)	(0.115)	(0.087)	(0.177)	(0.175)
Equiv. hh income (cube root)	-0.011**	0.030*	-0.003	0.002	-0.001	0.005	-0.004
	(0.004)	(0.012)	(0.012)	(0.008)	(0.005)	(0.010)	(0.005)
Constant	-9.187*	-17.679	-0.059	0.432	2.510	6.492	-4.023
	(4.345)	(10.819)	(4.064)	(2.540)	(1.525)	(4.365)	(3.622)
Wave dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	120	87	305	377	1,151	417	260
AIC	868	587	1,928	2,315	7,028	2,430	1,652
BIC	904	619	1,976	2,366	7,099	2,486	1,702

Table A.5, continued: Overestimating health and annual number of doctor visits at w + 1 by country

Note: The coefficients are based on the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.01

	(1)	(2)	(3)	(4)	(5)
	Main	Income 1	Income 2	Wealth	Alzheimer dropped
Health perception (ref.: concordance)					
Underestimating	0.244***	0.244***	0.244***	0.240***	0.243***
0	(0.018)	(0.018)	(0.018)	(0.018)	(0.019)
Chronic diseases	0.181***	0.181***	0.181***	0.179***	0.182***
	(0.005)	(0.005)	(0.005)	(0.005)	(0.005)
Activity limitations	0.096***	0.096***	0.096***	0.095***	0.097***
, ,	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)
Age	-0.001	-0.000	-0.001	0.004	-0.001
0	(0.011)	(0.011)	(0.011)	(0.011)	(0.011)
Age squared	0.000	0.000	0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Woman	0.042***	0.042***	0.042***	0.042***	0.042***
	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)
Educ. group (ref.: low)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)
Medium	0.006	0.008	0.005	0.018	0.007
hiculuit	(0.016)	(0.016)	(0.016)	(0.016)	(0.016)
High	-0.003	0.002	-0.004	0.020	-0.003
ingn	(0.018)	(0.002)	(0.001)	(0.019)	(0.019)
Retired	0.029	0.031	0.029	0.030	(0.019)
Rettied	(0.02)	(0.031)	(0.02)	(0.030)	(0.020)
Married	(0.017)	(0.017)	(0.017)	0.018	(0.017)
Married	(0.034)	(0.031)	(0.035)	(0.015)	-0.035
Equiv hh in come (subs reat)	0.001	(0.013)	(0.013)	(0.013)	(0.013)
Equiv. In mcome (cube root)	-0.001				-0.001
	(0.001)	0.005***	0.000***	0.00 0 ***	(0.001)
wave 5	-0.089	-0.085	-0.089	-0.092	-0.089
	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)
Equiv. hh income (cube root) 2		-0.002			
		(0.001)	0.000		
Equiv. hh income not normalised			-0.000		
			(0.000)		
Wealth quintile (ref.: 1st)					
2nd				-0.058**	
				(0.022)	
3rd				-0.103***	
				(0.021)	
4th				-0.110***	
				(0.022)	
5th				-0.130***	
				(0.022)	
Constant	1.507***	1.499***	1.485***	1.381***	1.491***
	(0.356)	(0.357)	(0.356)	(0.356)	(0.358)
Country dummies	Yes	Yes	Yes	Yes	Yes
N	46 067	46 067	46 067	46 067	45 917
AIC	260 957	260 954	260 958	260 875	259 975
BIC	261 202	260,204	260,200	260,075	260 219
1/1 5 .	L\\ I \ L\\ L	<u></u>	<u></u>	<u></u>	<u> </u>

Table A.6: Robustness analyses for annual doctor visits of the unimpaired sample

Note: The coefficients are based on the sample that is objectively unimpaired, i.e. able to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

	(1)	(2)	(3)	(4)	(5)
	Main	Income 1	Income 2	Wealth	Alzheimer dropped
Health perception (ref.: concordance)					
Overestimating	-0.146***	-0.146***	-0.146***	-0.145***	-0.144***
Ũ	(0.029)	(0.029)	(0.029)	(0.029)	(0.029)
Chronic diseases	0.149***	0.149***	0.149***	0.146***	0.152***
	(0.009)	(0.009)	(0.009)	(0.009)	(0.009)
Activity limitations	0.048***	0.048***	0.048***	0.047***	0.049***
5	(0.007)	(0.007)	(0.007)	(0.007)	(0.007)
Age	0.021	0.022	0.021	0.023	0.024
0	(0.018)	(0.018)	(0.018)	(0.019)	(0.018)
Age squared	-0.000	-0.000	-0.000	-0.000	-0.000
0 1	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Woman	0.014	0.014	0.014	0.015	0.009
	(0.028)	(0.028)	(0.028)	(0.028)	(0.028)
Educ. group (ref.: low)	()	()	()	()	()
Medium	-0.006	-0.005	-0.007	0.006	-0.007
	(0.033)	(0.033)	(0.033)	(0.033)	(0.033)
High	-0.087*	-0.084*	-0.088*	-0.069	-0.084*
	(0.042)	(0.043)	(0.042)	(0.043)	(0.042)
Retired	0.014	0.016	0.014	0.019	0.011
Remeu	(0.033)	(0.033)	(0.033)	(0.033)	(0.033)
Married	0.019	0.023	0.019	0.034	0.012
munica	(0.030)	(0.031)	(0.030)	(0.030)	(0.030)
Equiv bh income (cube root)	-0.001	(0.001)	(0.000)	(0.000)	-0.001
Equiv. Int meonie (cube foot)	(0.001)				(0.001)
Wave 5	-0.046	-0.042	-0.046	-0.047	-0.045
Wave 5	(0.038)	(0.038)	(0.038)	(0.038)	(0.038)
Equiv bh income (cube root) ?	(0.050)	0.001	(0.050)	(0.050)	(0.050)
Equiv. In meome (cube root) 2		(0.001)			
Equiv bh income not normalised		(0.002)	0.000		
Equiv. In income not normansed			-0.000		
Wealth quintile (raf. 1at)			(0.000)		
and and an				0.050	
2110				-0.030	
2				(0.036)	
Sra				-0.110^{-1}	
446				(0.039)	
4th				-0.119***	
Eth.				(0.040)	
Sth				-0.086	
Grandrad	1 404*	1 404*	1 401*	(0.045)	1 000*
Constant	1.434*	1.434*	1.421*	1.382	1.338*
	(0.646)	(0.645)	(0.643)	(0.645)	(0.645)
Country dummies	Yes	Yes	Yes	Yes	Yes
N	7,801	7,801	7,801	7,801	7,709
AIC	50,545	50,544	50,545	50,534	49,893
BIC	50.740	50,739	50,740	50,749	50.088

Table A.7: Robustness analyses for annual doctor visits of the impaired sample

Note: The coefficients are based on the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

	(1) Objectively	(2) Objectively	(3) Objectively	(4) Objectively	(5) Objectively	(6) Objectively
	Unimpaired	Unimpaired	Unimpaired	Impaired	Impaired	Impaired
	Main	Access	Insurance	Main	Access	Insurance
Health perception (ref.: concordance)	0.0=0***	0.0(1***	0.050***			
Underestimating	0.259	0.261	0.259			
	(0.021)	(0.022)	(0.021)	0 4 40***	0 4 4 0 * * *	0 4 40***
Chronic diseases	0.176***	0.173***	0.176***	0.143***	0.140***	0.143***
	(0.006)	(0.006)	(0.006)	(0.010)	(0.011)	(0.010)
Activity limitations	0.093***	0.095***	0.094^{***}	0.048^{***}	0.052***	0.049***
	(0.011)	(0.011)	(0.011)	(0.007)	(0.007)	(0.007)
Age	0.004	0.006	0.003	0.028	0.026	0.028
	(0.012)	(0.012)	(0.012)	(0.021)	(0.022)	(0.021)
Age squared	0.000	0.000	0.000	-0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Woman	0.038**	0.038**	0.039**	-0.036	-0.034	-0.036
	(0.014)	(0.015)	(0.014)	(0.032)	(0.033)	(0.032)
Educ. group (ref.: low)						
Medium	0.022	0.026	0.022	-0.005	-0.003	-0.006
	(0.018)	(0.018)	(0.018)	(0.036)	(0.038)	(0.036)
High	0.007	0.009	0.004	-0.091	-0.080	-0.089
0	(0.021)	(0.022)	(0.021)	(0.048)	(0.049)	(0.048)
Retired	0.026	0.025	0.026	-0.012	-0.023	-0.013
	(0.021)	(0.021)	(0.021)	(0.038)	(0.040)	(0.038)
Married	-0.037*	-0.040*	-0.038*	0.046	0.055	0.046
	(0.016)	(0.017)	(0.016)	(0.031)	(0.032)	(0.031)
Fauiv hh income (cube root)	-0.001	-0.001	-0.001	-0.001	-0.001	-0.001
Equiv. Int meonie (cube 1001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Health access	(0.001)	0.020	(0.001)	(0.002)	0.021	(0.002)
Ticului access		(0.023)			(0.033)	
Supplementary incurance		(0.025)	0.019		(0.000)	-0.046
Supplementary insurance			(0.01)			-0.040
Overestimating			(0.021)	0 147***	0 149***	(0.030)
Overestillating				(0.022)	-0.142	(0.022)
Constant	1 20/**	1 00/**	1 22 4***	(0.032)	(0.033)	(0.032)
Constant	1.500	1.220	1.524	(0.714)	1.105	1.1/1
	(0.398)	(0.407)	(0.398)	(0.714)	(0.755)	(0.715)
Country dummies	Yes	Yes	Yes	Yes	Yes	Yes
N	33,984	32,023	33,921	5,840	5,417	5,812
AIC	192,107	180,867	191,768	37,858	35,136	37,681
BIC	192,317	181,084	191,987	38,025	35,308	37,855

Table A.8: Robustness analyses for Wave 5

Note: "Unimpaired" refers to the sample that is objectively unimpaired, i.e. able to stand up from the chair and "Impaired" refers to the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 6. All explanatory variables are taken from Wave 5. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

	(1)	(2)	(2)	()
	(1)	(2)	(3)	(4)
	Objectively	Objectively	Objectively	Objectively
	Unimpaired	Unimpaired	Impaired	Impaired
	Wave 2	Wave 5	Wave 2	Wave 5
Health perception (ref.: concordance)				
Underestimating	0.189***	0.259***		
	(0.037)	(0.021)		
Overestimating			-0.129*	-0.147***
			(0.062)	(0.032)
Chronic diseases	0.201***	0.176***	0.166***	0.143***
	(0.011)	(0.006)	(0.022)	(0.010)
Activity limitations	0.114^{***}	0.093***	0.045^{*}	0.048^{***}
	(0.021)	(0.011)	(0.019)	(0.007)
Age	-0.077*	0.004	0.011	0.028
-	(0.036)	(0.012)	(0.076)	(0.021)
Age squared	0.001^{*}	0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.001)	(0.000)
Woman	0.050*	0.038**	0.173***	-0.036
	(0.023)	(0.014)	(0.052)	(0.032)
Educ. group (ref.: low)				
Medium	-0.048	0.022	-0.014	-0.005
	(0.029)	(0.018)	(0.067)	(0.036)
High	-0.032	0.007	-0.065	-0.091
-	(0.033)	(0.021)	(0.079)	(0.048)
Retired	0.023	0.026	0.067	-0.012
	(0.029)	(0.021)	(0.058)	(0.038)
Married	-0.012	-0.037*	-0.103	0.046
	(0.028)	(0.016)	(0.071)	(0.031)
Equiv. hh income (cube root)	-0.002	-0.001	-0.002	-0.001
	(0.001)	(0.001)	(0.003)	(0.002)
Constant	3.622**	1.306**	1.876	1.168
	(1.118)	(0.398)	(2.440)	(0.714)
Country dummies	Yes	Yes	Yes	Yes
Ν	12,083	33,984	1,961	5,840
AIC	68,559	192,107	12,603	37,858
BIC	68,736	192,317	12,737	38,025

Table A.9: Robustness analysis: Annual number of doctor visits at w + 1 by survey wave

_

_

Note: "Unimpaired" refers to the sample that is objectively unimpaired, i.e. able to stand up from the chair and "Impaired" refers to the sample that is objectively impaired, i.e. unable to stand up from the chair. The dependent variable "doctor visits" is based on the annual number of doctor visits, visits to emergency rooms and outpatient clinic visits at wave w + 1, i.e. Wave 4 or Wave 6. All explanatory variables are taken from wave w, i.e. Wave 2 or Wave 5 respectively. The estimated coefficients are based on a negative binomial regression model with mean dispersion. Standard errors are clustered at the household level and presented in parentheses. * p<0.05, ** p<0.01, *** p<0.001

VIENNA INSTITUTE OF DEMOGRAPHY

Working Papers

Kebede, Endale, The Relative Importance of Female Education on Fertility Desires in Sub-Saharan Africa: A Multi-Level Analysis, VID Working Paper 09/2019.

Winkler-Dworak, Maria, Eva Beaujouan, Paola Di Giulio and Martin Spielauer, *Simulating Family Life Courses: An Application for Italy, Great Britain, and Scandinavia,* VID Working Paper 08/2019.

Hammer, Bernhard, Sonja Spitzer, Lili Vargha and Tanja Istenič, *The Gender Dimension* of Intergenerational Transfers in Europe, VID Working Paper 07/2019.

Wazir, Muhammad Asif and Anne Goujon, *Assessing the 2017 Census of Pakistan Using Demographic Analysis: A Sub-National Perspective*, VID Working Paper 06/2019.

Yegorov, Yuri, Dieter Grass, Magda Mirescu, Gustav Feichtinger and Franz Wirl, *Growth and Collapse of Empires: A dynamic Optimization Model*, VID Working Paper 05/2019.

Chisumpa, Vesper H., Clifford O. Odimegwu and Nandita Saikia, *Mortality in Sub-Saharan Africa: What is Killing Adults Aged 15-59 Years in Zambia?*, VID Working Paper 04/2019.

Striessnig, Erich and Jayanta Kumar Bora, *Under-Five Child Growth and Nutrition Status: Spatial Clustering of Indian Districts*, VID Working Paper 03/2019.

Speringer, Markus, Anne Goujon, Samir K.C., Michaela Potančoková, Claudia Reiter, Sandra Jurasszovich and Jakob Eder, *Global Reconstruction of Educational Attainment*, 1950 to 2015: *Methodology and Assessment*, VID Working Paper 02/2019.

Testa, Maria Rita and Danilo Bolano, *Intentions and Childbearing in a Cross-Domain Life Course Approach: The Case of Australia*, VID Working Paper 01/2019.

Goujon, Anne, Claudia Reiter and Michaela Potančoková, *Religious Affiliations in Austria at the Provincial Level: Estimates for Vorarlberg*, 2001-2018, VID Working Paper 13/2018.

Spitzer, Sonja, Angela Greulich and Bernhard Hammer, *The Subjective Cost of Young Children: A European Comparison*, VID Working Paper 12/2018.

Jatrana, Santosh, Ken Richardson and Samba Siva Rao Pasupuleti, *The Effect of Nativity*, *Duration of Residence, and Age at Arrival on Obesity: Evidence from an Australian Longitudinal Study*, VID Working Paper 11/2018.

The Vienna Institute of Demography Working Paper Series receives only limited review. Views or opinions expressed herein are entirely those of the authors.